Is minimum-variance investing really worth the while? An analysis with robust performance inference

Patrick Behr^{*} André Güttler[†] Felix Miebs[‡]

October 31, 2008

^{*}Department of Finance, Goethe-University Frankfurt, Mertonstr. 17, 60054 Frankfurt, Germany, E-mail: behr@finance.uni-frankfurt.de.

[†]HCI Endowed Chair of Financial Services, Department of Finance, Accounting and Real Estate, European Business School, International University, Rheingaustr. 1, 65375 Oestrich-Winkel, Germany, E-mail: andre.guettler@ebs.edu(Corresponding author).

[‡]HCI Endowed Chair of Financial Services, Department of Finance, Accounting and Real Estate, European Business School, International University, Rheingaustr. 1, 65375 Oestrich-Winkel, Germany, E-mail: felix.miebs@ebs.edu

Is minimum-variance investing really worth the while? An analysis with robust performance inference

Abstract

This paper examines the risk-adjusted performance of the minimum-variance equity investment strategy in the U.S. While earlier studies only relied on empirical Sharpe ratio comparisons between the (constrained) minimum-variance strategy and different benchmark portfolios, we employ bootstrap methods for statistical inference concerning the Sharpe ratio, the Sortino ratio, certainty equivalents and alpha measures based on several factor models. We confirm and provide robust inference concerning earlier findings that constrained minimum-variance portfolios do outperform a value weighted benchmark. Moreover, our findings are in line with prior research, stating that minimum-variance portfolios do not outperform a naively diversified benchmark in terms of the Sharpe ratio. Both our results are invariant to the portfolio revision frequency and may be observed in all subperiods. Nevertheless, we show the high sensitivity of the constrained minimum-variance portfolios to the revision frequency and the imposed maximum portfolio weight constraints.

1 Introduction

The foundations of modern portfolio theory go back to the seminal work of Markowitz (1952; 1959). In his framework, portfolio selection is postulated as a trade-off between risk and return, where efficient portfolios deliver a maximum return for a given level of risk or, vice versa, deliver a minimum of risk for a given level of return. Ever since then a vast amount of literature has been devoted to research about modern portfolio selection. A considerable amount of this research has focused on the evaluation of the out-of-sample performance of two portfolios on the efficient frontier, which comprises all efficient portfolios, namely the tangency and the minimum-variance portfolio.

The tangency portfolio, as the portfolio with the largest excess return per unit of risk, attracted researchers' attention due to its theoretical foundation as the optimal portfolio for an investor. Following Tobin's mutual fund separation (1958), every investor should hold (dependent on her risk aversion) a certain fraction of wealth in the tangency portfolio, but should not alter the weight of any asset in the tangency portfolio.

Stein (1956), Frost and Savarino (1986), Jorion (1986) and Black and Litterman (1992), amongst others, point out that the estimation of expected returns (from sample data), which is necessary for the calculation of the tangency portfolio, is error prone and may yield misleading results. Additionally, Goyal and Welch (2003a; 2007) and Butler et al. (2001) show that even more advanced techniques for the return prediction, based on predictive regressions, have a similarly poor out-of-sample performance. In line with this, Bloomfield et al. (1977), Jobson and Korkie (1981a) and Jagannathan and Ma (2003) find that the tangency portfolio does not outperform an equally weighted portfolio.

These sobering results concerning the performance of the tangency portfolio drew attention to the minimum-variance portfolio, the only portfolio on the efficient frontier that requires the variance-covariance matrix as input parameter for the optimization. Merton (1980), Jorion (1985), and Nelson (1992) point out that variance-covariance estimates are relatively stable over time and can, hence, be predicted more reliably than returns. Underpinning these results of lower estimation errors for the minimum-variance portfolio, Baker and Haugen (1991) and Clarke et al. (2006) find an out-of-sample outperformance of the minimum-variance portfolio relatively to a value weighted portfolio. Additionally, Chan et al. (1999), Jagannathan and Ma (2003) and DeMiguel et al. (2007) point out that the short-sale constrained minimum-variance portfolio outperforms the tangency portfolio approach.

Evidence regarding a possible outperformance of the minimum-variance portfolio relative to an equally weighted portfolio is less clear. Based on purely descriptive results, Chan et al. (1999) and Jagannathan and Ma (2003) report higher out-of-sample Sharpe ratios for the constrained minimum-variance portfolio (CMVP). Given their lack of statistical inference, a robust conclusion whether the (constrained) minimum-variance strategy offers superior risk-adjusted returns cannot be made. Employing a parametric test, DeMiguel et al. (2007) find statistically indistinguishable differences in Sharpe ratios between the minimum-variance portfolio and a value weighted and equally weighted portfolio. Given this mixed evidence, the question whether minimum-variance investing is worthwhile appears to be rather unacknowledged so far.

The aim of this paper is to assess the risk-adjusted performance of the minimum-variance strategy for equity investors using robust performance inference. Though the research question, whether minimum-variance investing can deliver a risk adjusted outperformance relatively to a value or equally weighted benchmark strategy, has already by een investigated, we offer several new results and perspectives.

First, all aforementioned papers failed to provide robust evidence whether the minimumvariance strategy is worthwhile to conduct. This paper is, to the best of our knowledge, the first to use nonparametric performance tests between the minimum-variance investment strategy employed and a naively diversified as well as a value weighted benchmark portfolio. The employed nonparametric bootstrap approach should be better suited for this kind of performance comparison since neither the strong assumption of normally distributed nor independent and identically distributed (i.i.d.) returns are required.¹ Furthermore, we provide beside the Sharpe ratio performance comparisons for the Sortino ratio, alpha measures based on diverse factor models and certainty equivalents in order to provide robust evidence of possible performance differences.

Second, we are the first to base the derived results on the whole U.S. equity universe as captured by the Center for Research in Security Prices (CRSP) stock database², which allows a comprehensive assessment of the considered research question. Contrary, all

¹Evidence for non-normality of stock returns goes back to Fama (1965). Additionally, DeMiguel et al. (2007) note that "this assumption is typically violated in the data". Ledoit and Wolf (2008) shows that the commonly used parametric Sharpe ratio test by Jobson and Korkie (1981a) requires, likewise the corrected version of the test by Memmel (2003), identical and independently distributed returns.

²For a closer description of the dataset see section 3.

aforementioned papers base their findings on either randomly drawn samples from the CRSP database (e.g. Chan 1999 and Jagannathan and Ma 2003), stock index constituents (e.g. Baker and Haugen 1991 and Clarke et al. 2006) or sector portfolios (DeMiguel et al. 2007).

During the course of our paper, we reproduce with our dataset a part of the work by DeMiguel et al. (2007) - in particular by using a maximum portfolio weight³ of 2% in the portfolio optimization process. Despite striking differences in the inference methodology⁴ and different datasets, we find evidence in line with DeMiguel et al. (2007) that the 2% CMVP offers no outperformance compared to naively diversified portfolios. Thus, in this setup, we do not corroborate (descriptive) findings by Chan et al. (1999) and Jagannathan and Ma (2003). Nevertheless, we do find evidence, by relaxing the maximum portfolio weight constraint, that CMVPs deliver a favorable performance in terms of alpha based on every considered factor model in comparison to the equally weighted benchmark portfolio. Performance comparisons relative to the value weighted benchmark portfolio.

The remainder of the paper is organized as follows: section two reviews the methodology and gives a thorough outline of the employed nonparametric performance tests. Section three describes the data used. Section four reports and discusses the empirical results whereas section five presents robustness checks. Finally, the last section concludes the findings of this paper.

2 Methodology

This section is divided into three parts. First, we describe the portfolio optimization methodology used for the derivation of the CMVPs. This is followed by a short review of the various performance measures, while the third part of this section provides a thorough outline of the employed nonparametric bootstraps for the statistical inference concerning the comparison of the considered portfolio performance metrics.

³In the following we use the terms "maximum portfolio weight", "upper bound" and "upper constraint" synonymously.

⁴DeMiguel et al. (2007) use the Jobson and Korkie (1981a) parametric Sharpe ratio test in its corrected version by Memmel (2003), while we employ a nonparametric bootstrap for inference concerning the equality of Sharpe ratios of the minimum-variance portfolio and the respective benchmarks.

2.1 Portfolio optimization

Despite the fact that estimates of the variance-covariance matrix are less error prone than those of expected returns, the estimation error problem is far from alleviation. Actually, Chan et al. (1999) point out that the estimation error in variance-covariance estimates from sample data is still substantial. For example, they find a correlation of only 34% (18%) between the in-sample variance-covariance matrix and the out-of-sample covariance matrix, based on a 36 (12) months realization period.

There are various ways to deal with estimation errors in order to improve the out-ofsample reliability of parameter estimates. In general, three major approaches can be distinguished: factor model approaches, Bayes-Stein shrinkage estimators and the implementation of portfolio weight constraints.⁵

Starting with the postulation of Sharpe's (1963; 1964) one factor model, the use of factor models, not only to explain and predict returns, but also to estimate the variancecovariance matrix by imposing the factor model structure on the estimator of the variancecovariance matrix, became popular. The basic assumption underlying the one factor model in the context of covariance prediction is that the mutual return correlations between all assets is attributable to a common factor - the market index. Chan et al. (1999) provide a comprehensive examination of various factor models and their capability of out-of-sample variance-covariance prediction. They show, however, that the forecasted variance-covariance matrix based on historical returns performs for the purpose of portfolio optimization about as well as factor model estimates.

The most prominent approach in dealing with the estimation error problem is probably the Bayes-Stein shrinkage estimator. Initially proposed by Stein (1956) and James and Stein (1961), Bayes-Stein shrinkage estimators have widely been applied to the estimation of both input parameters of portfolio optimization, expected returns⁶ and the variancecovariance matrix⁷. However, Jobson and Korkie (1980) show that shrinkage estimators do not perform well in small samples. Additionally, Jagannathan and Ma (2003) reveal that the Bayes-Stein improvements in the variance-covariance matrix are minor relative

⁵Other solutions include equilibrium constraints (Black and Litterman 1992), optimal combinations of portfolios (Garlappi et al. 2007) and the formulation of robust portfolio optimization problems, which directly incorporate a measure of the parameter uncertainty into the optimization procedure. Fabozzi et al. (2007) provide a sound survey of the latter approach.

⁶See for instance Jobson et al. (1979), Jobson and Korkie (1981a) and Jorion (1985; 1986)

⁷We refer to Ledoit and Wolf (2003) for an application of shrinkage techniques to the variance-covariance matrix.

to estimates based on historical data. While DeMiguel et al. (2007) confirm this finding for both, the variance-covariance matrix and expected returns.

The implementation of portfolio weight constraints in order to avoid highly concentrated portfolios and extreme portfolio positions, that are frequently derived in the mean-variance optimization procedure⁸, has, to the best of our knowledge, first been applied by Frost and Savarino (1988). Nevertheless, Jagannathan and Ma (2003) were the first who studied the shrinkage like effect of upper and lower (short-sale constraint) bounds on portfolio weights, preventing large exposure to single assets. Despite their promising findings only little attention has been payed to the proposed adjustments since then. However, DeMiguel et al. (2007) recently showed that the minimum-variance portfolio with the proposed portfolio weight restrictions delivers the most favorable out-of-sample performance in terms of the Sharpe ratio.

In line with DeMiguel et al. (2007) and Chan et al. (1999) we adopt the constrained myopic minimum-variance portfolio setup proposed by Jagannathan and Ma (2003) that builds on the Markowitz (1952) portfolio selection framework. Since the crucial assumption of normally distributed returns is typically violated in the data.⁹ we meet the assumption that the investor's utility function is quadratic in order to keep the portfolio optimization approach valid.¹⁰ Thus, we employ the CMVP optimization setup on a "rolling-sample" basis. Since the choice of sample size for the parameter estimation represents a delicate trade-off between an increase of statistical confidence and the incorporation of possibly irrelevant data¹¹, we stick to the sample size choice of Chan (1999) and Jagannathan and Ma (2003). Accordingly, the portfolio is rebalanced every three months in the base case, taking the return observations of the last 60 months as input parameter for the variance-covariance matrix estimation.¹²

Hence, we solve the standard myopic portfolio optimization problem, imposing the recommended constraints by Jagannathan and Ma (2003) in order to reduce the estimation error and to achieve a shrinkage-like effect.¹³ Accordingly, we set on the portfolio weights

⁸See Green and Hollifield (1992), Chopra (1993) and Chopra and Ziemba (1993) for evidence concerning this point.

 $^{^{9}}$ For evidence concerning this point see Fama (1965) and DeMiguel et al. (2007)

¹⁰All portfolio optimization approaches that base on Markowitz (1952) require either normally distributed returns or quadratic utility functions of investors.

¹¹See for this Jobson (1981b) and Levine (1972).

¹²Additionally, we vary the portfolio revision frequency for robustness check purposes from three months to six and twelve months. Results for this are reported in section 5.

¹³Even though the setup is multi-period in nature, Mossin (1969), Fama (1970), Hakansson (1970; 1974) and Merton (1990) show that "under several sets of reasonable assumptions, the multi-period problem

a no-short sales constraint, $w_{i,t} \ge 0$ and an upper bound w^{max} , which is varied in one percent steps in the range $\{w^{max} = 2\%, 3\%, ..., 20\%\}$, on the weight a single stock may have in the portfolio. After the optimization process the portfolio weights remain three months¹⁴ unchanged up to the next optimization. Formally, this results in an optimization problem for each period t of the following form¹⁵:

$$\mathbf{\mathbf{w}_{t}^{\min} \mathbf{w}_{t}^{'} S \mathbf{w}_{t} s.t.}_{N}$$
(1)

$$\sum_{i=1}^{n} \mathbf{w}_{i,t} = 1 \tag{2}$$

$$\mathbf{w}_{i,t} \ge 0, \text{ for } i = 1, 2, ..., N$$
 (3)

 $\mathbf{w_{i,t}} \le w^{max}, \text{ for } i = 1, 2, \dots, N \tag{4}$

The Kuhn-Tucker conditions (necessary and sufficient) are accordingly:

$$\sum_{i} \mathbf{S}_{\mathbf{i},\mathbf{j}} \mathbf{w}_{\mathbf{j}} - \lambda_{\mathbf{i}} + \delta_{\mathbf{i}} = \lambda_0 \ge 0, \text{ for } i = 1, 2, ..., N$$
(5)

$$\lambda_{\mathbf{i}} \ge 0 \text{ and } \lambda_{\mathbf{i}} = 0 \text{ if } \mathbf{w}_{\mathbf{i}} > 0, \text{ for } i = 1, 2, ..., N$$
 (6)

$$\delta_{\mathbf{i}} \ge 0 \text{ and } \delta_{\mathbf{i}} = 0 \text{ if } \mathbf{w}_{\mathbf{i}} < w^{max}, \text{ for } i = 1, 2, \dots, N$$

$$(7)$$

The notation hereby is as follows: **w** denotes the vector of portfolio weights, **S** the empirically estimated variance-covariance matrix, λ the vector of Lagrange multipliers in the non negativity constraint of portfolio weights, δ the multipliers of upper bound portfolio weight constraints and λ_0 the multiplier for the portfolio weights to sum up to one.

2.2 Performance metrics

In a second step we evaluate the out-of-sample performance of the CMVP and compare it to both, a value weighted benchmark portfolio, which serves as a market proxy, and an equally weighted portfolio, which is considered as a simple benchmark asset allocation

can be solved as a sequence of single-period problems" (Elton and Gruber 1997).

¹⁴The portfolio revision frequency is varied for robustness check pruposes to six and twelve months. Results for these revision frequencies are reported in section 5.2.

¹⁵The notation and variable explanation follows closely Jagannathan (2003). Bold factors indicate vectors and matrices, whereas e.g. $\mathbf{w_i}$ denotes the i-th element of vector \mathbf{w} .

strategy following for instance Bloomfield et al. (1977) and DeMiguel et al. (2007).¹⁶ In order to provide a broad picture of risk adjusted portfolio performance, we report various performance measures which incorporate different conceptual measures of risk. Accordingly, we provide evidence based on the certainty equivalent (CEQ)¹⁷, the Sharpe (SH) and Sortino (SR) ratio as well as on alpha measures based on a one factor model (as initially proposed by Jensen 1969), the Fama and French (1992) three factor and Carhart (1997) four factor model, which are given by

$$SH = \frac{r_p - r_f}{\sigma_p} \tag{8}$$

$$SR = \frac{r_p - r_f}{\sigma_p^d} \tag{9}$$

$$CEQ = r_p - \frac{\dot{\gamma}}{2}\sigma_p^2 \tag{10}$$

$$\alpha_p = r_{p,t} - \sum_{k=1}^{K} \beta_{p,k} r_{k,t} - \epsilon_{p,t}, \text{ with } \epsilon \sim F_p(0,\sigma^2)$$
(11)

Thereby denotes r_p the return, σ_p the standard deviation and σ_p^d the downside deviation¹⁸ of portfolio p, while α_p stands for the portfolio specific constant, factor independent, return, $\sum_{k=1}^{K} \beta_{p,k}$ the sensitivities of the portfolio return relative to the return of the K explanatory factors $r_{k,t}$ and $\epsilon_{p,t}$ the portfolio specific white noise error term.

2.3 Statistical performance metric inference

In order to check upon the statistical significance between the derived performance measures of the CMVPs and the considered benchmarks, we employ nonparametric bootstraps for a robust performance measure inference. This is noteworthy since none of the earlier mentioned papers employed a robust inference methodology for the comparison between the (constrained) minimum-variance strategy and alternative investment strategies. An exception is the work by DeMiguel et al. (2007), who employed a parametric Sharpe

¹⁶For convenience and following the common acceptance of the CRSP value and CRSP equally weighted market indices as market proxies (see Lehmann and Modest 1987) we opted to choose both as benchmark portfolios.

¹⁷We assume that investors have quadratic utility functions which commonly accepted (see e.g. DeMiguel et al. 2007). We choose to set the risk aversion parameter to $\gamma = 4$ in order to assess the profitability of the CMVPs from the point of view of a highly risk averse investor

¹⁸For the purpose of our paper, we set the shortfall threshold to zero, in order to have a common, non portfolio specific shortfall threshold for all portfolios in the analysis.

ratio test by Jobson and Korkie (1981b)¹⁹. Nevertheless, it is important to point out that the parametric Sharpe ratio test by Jobson and Korkie (1981b) (as well as in its corrected version by Memmel 2003) requires the strong assumption of normally and i.i.d. distributed portfolio returns.²⁰ The effect of non-normality of returns on this test and the resulting inappropriateness is discussed by Ledoit and Wolf (2008).

We have therefore opted to use a nonparametric bootstrap approach, which was initially proposed by Efron (1979). The superiority of the bootstrap approach vis-à-vis parametric tests is twofold. First, it does not require any assumption about the distribution of the considered performance measures nor their differences. Instead, our approach allows ous to draw inference from our sample distribution. Second, Navidi (1989) showed that the bootstrap is under all circumstances at least as good as the normal approximation which is required for the widely used test by Jobson and Korkie (1981b). The use of bootstrap methodology for robust performance metric inference has been highlighted by Morey and Vinod (1999) and Kosowski et al. (2004; 2007) with regard to the Sharpe ratio, using a simple observation bootstrap, and the factor model based alpha measures, using a residual bootstrap respectively.

2.3.1 Observation bootstrap

Our approach towards testing for the equality of performance measures incorporating a total measure of risk, namely the CEQ, the Sharpe and Sortino ratios follows the work by Morey and Vinod (1999) and can be summarized as follows: From the empirical sample comprising N monthly returns we draw a random return $r_{p,t}$ from the time series of portfolio p with replacement. It is important to note that this is done pair wise in a timely fashion, meaning that the return of the same randomly selected month is chosen from the respective time series of the benchmark portfolio returns $r_{bm,t}$. This is repeated n times, leaving a bootstrap sample of n returns for each CMVP p { $r_{p,t}$ with t = 1, 2, ..., n} and benchmark portfolio bm { $r_{bm,t}$ with t = 1, 2, ..., n}. Repeating the prior step B times yields B bootstrap samples for every CMVP p { S_p^b with b = 1, 2, ..., B} and benchmark portfolio bm { S_{bm}^b with b = 1, 2, ..., B}, each containing n returns. For each of the B bootstrap samples we compute the Sharpe and Sortino ratio differences between each

¹⁹Memmel (2003) proposed a minor correction for this test statistic.

²⁰Evidence concerning the non-normal distribution of the CMVP returns (at least for our results) is provided in the tables 3 and 8.

particular CMVP and the (naively or the value weighted) benchmark portfolios. Sorting each of the resulting B CEQ, Sharpe and Sortino ratios differences in a vector enables us to construct confidence intervals and to conduct hypothesis testing.

Point of interest is the confidence level at which the CMVP offers a statistically significant outperformance relative to the considered benchmark portfolios. Accordingly, we compute the p-values for the one sided hypotheses that the difference between a particular portfolio performance measure of the CMVP and the benchmark portfolio (either equally or value weighted) is smaller or equal to zero:

$$H_0: \widehat{CEQ}_{min-var} - \widehat{CEQ}_{benchmark} \le 0 \tag{12}$$

$$H_0: \widehat{SH}_{min-var} - \widehat{SH}_{benchmark} \le 0 \tag{13}$$

$$H_0: \widehat{SR}_{min-var} - \widehat{SR}_{benchmark} \le 0 \tag{14}$$

The p-values are thereby computed by finding the element entry number of the first observation in each particular vector of sorted bootstrap differences, which has a non-negative sign. This number is in turn divided by the number of bootstrap iterations, which then yields the desired p-value for the one sided hypothesis test that the respective performance metric of the CMVP exceeds that of the benchmark portfolio. Stated differently, the p-value delivers the fraction of bootstrap iterations, and accordingly the probability, in favor of a rejection of the null hypothesis.

2.3.2 Residual bootstrap

To draw robust performance inference using factor model approaches, we adopt the residual bootstrap methodology proposed by Kosowski et al. (2004; 2007). The basic idea is that the true return data generating process P of a portfolio p is fully described by a K-factor model as given provided in equation (11). Following this, we try to capture the underlying data generating process for each portfolio P_p from sample data. This estimate, denoted by $\hat{P}_p = (\hat{\alpha}_p \ \hat{\beta}_p, F_{p,e})$, with $F_{p,e}$ being the empirical cumulative density function of the residuals $\hat{\epsilon}_p$, serves in the following as the data generating process for the creation of the B bootstrap samples.

Thus, we estimate in a first step for each of the CMVPs p the respective data generating

process, \hat{P}_p , from the sample data. Following this, we store the estimated parameter values of $\hat{\alpha}_p$, $\hat{\beta}_p$, the t-value of $\hat{\alpha}_p$, as well as the time series of the estimated residuals $\hat{\epsilon}_{p,t}$ with t=1,2,...,T.

In a second step we construct an artificial time series of portfolio returns, r_p^b , which has an intercept term of zero by construction. This is done for any particular CMVP p by multiplying the estimated sensitivities to the considered K explanatory factors $\sum_{k=1}^{K} \beta_{p,k}$ with the respective factor returns r_k with k=1,2,...,K. Additionally, a bootstrapped residual from the empirical density function of the residuals, \hat{e}_p^b , is added. This operation should not alter the intercept of the regression line, since the residuals, \hat{e}_p , are white noise with zero mean. Accordingly, the regression line should, by construction, still pass the origin, which would be reflected by an alpha measure of zero. Formally, the constructed artificial time series for each portfolio p for which a K-factor model has been considered to capture the return data generating process is given by:

$$r_{p,t}^{b} = \sum_{k=1}^{K} \hat{\beta}_{p,k} r_{k,t} + \hat{\epsilon}_{p,t}^{b} \text{ with } t=1,2,...,T$$
(15)

In the following final step we re-estimate for every portfolio p all regression parameters in equation (11) from the artificial time series:

$$r_{p,t}^{b} = \hat{\alpha}_{p}^{b} + \sum_{k=1}^{K} \hat{\beta}_{p,k}^{b} r_{k,t}$$
(16)

The resulting intercept coefficient $\hat{\alpha}_p^b$ is now of special interest, since the sampling variation, reflected in the bootstrapped residuals, should, if the empirical parameter, $\hat{\alpha}_p$ were significant, not yield a parameter value as high as the empirical value. Repeating the prior steps b = 1, 2, 3, ..., B times yields the desired number of B bootstrap samples and corresponding B parameter estimates. Sorting these parameters and applying the same test procedure as for the observation bootstrap leaves ous with the following hypothesis:

$$H_0: \hat{\alpha} \le \hat{\alpha}^b \tag{17}$$

Since the t-value has more favorable statistical properties, due to the additional coverage of the estimation precision of $\hat{\alpha}$, Kosowski et al. (2004; 2007) propose to bootstrap the t-value.²¹ Following this argument, the resulting one sided hypothesis to test for is given by:

$$H_0: t_{\hat{\alpha}} \le t_{\hat{\alpha}^b} \tag{18}$$

3 Data

Our dataset comprises the entire CRSP monthly stock database from April 1964 to December 2007. From this database, we only consider stocks in the optimization that have non-zero returns and no missing values in the variance-covariance estimation period of 60 months prior to the optimization's point in time. It is important to stress the necessity of this filter since thin trading, reflected in limited or even no trading of stocks, drives the covariance of these illiquid stocks with other stocks towards zero. In consequence of the resulting downward biased covariance, foremost illiquid stocks will be selected so that the performance will likely be driven by soaking up the stock market inherent liquidity premium²². In order to avoid these thin trading effects, the aforementioned filter is employed.²³ In case of missing values in the out-of-sample period we opted to set the stock return to zero.

Finally, we follow the argument by Chan et al. (1999), who claim that the variancecovariance matrix becomes too noisy if very small and micro capitalized firms are included in the sample. Hence, we include in our final sample only those stocks that are in the upper 80% size percentile (according to their market capitalization) and have a stock price of more than \$5.

Time series data for interest rates and factor return data for the factor models described

²¹We control for autocorrelation and heteroscedasticity using the autocorrelation and heteroscedasticity consistent covariance matrix from Newey and West (1987).

 $^{^{22}}$ For evidence concerning the liquidity premium inherent in stock markets see Pastor and Stambaugh (2003).

²³Despite the availability of adjustment procedures for the thin trading effect concerning the covariance estimation (see e.g. Scholes and Williams 1977 and Dimson 1979), none of these adjustments has proven to account properly for the thin trading effect. Evidence for this is provided by a comprehensive study by McInish and Wood (1986).

in section 2.2 are obtained from Kenneth French's website²⁴, while the National Bureau of Economic Research (NBER) recession period data, which we use for the determination of U.S. recession periods²⁵ is taken from the corresponding NBER website²⁶.

4 Empirical results

The first part of this section describes the empirical performance measures of the CMVPs with a portfolio revision frequency of three months over the complete sample period from April 1968 to December 2007, while the second part provides statistical inference for the empirical results and checks upon the significance of the empirical findings.

4.1 Descriptive performance metric analysis

Since the minimum-variance approach aims at the reduction of risk it is suggestive to assess the empirical standard deviation of the CMVPs first. Quite obvious is the almost nondecreasing pattern of standard deviations for the less CMVPs, which is accompanied by a decreasing average number of stocks in the respective minimum-variance portfolios. Accordingly, the observable increase in standard deviations can be attributed to an increase in idiosyncratic risk, due to the reduced deterministic diversification of less restrictively CMVPs. This effect is reflected in the declining values of the adjusted R^2 as depicted in tables 1 and 2, which is observable for every considered benchmark portfolio and factor model. Noteworthy is however the reduced standard deviation of the CMVPs. As depicted in table 3 the CMVPs deliver, up to a portfolio weight constraint of 8% (19%), standard deviations below those of the value (equally) weighted benchmark portfolio. Nevertheless, this risk reduction bares the "cost" of lower returns.²⁷

A quantification of the observable trade-off between risk reduction and return is given by the described performance metrics in section 2.2. The broad picture in table 3 shows that all CMVPs clearly outperform the value weighted benchmark based on all performance metrics that incorporate a total measure of risk. This is especially interesting for

²⁴http://mba.tuck.dartmouth.edu/pages/faculty/ken.french/data_library.html

 $^{^{25}}$ We are in line with other studies (e.g. Campbell et al. 2001 and Goyal and Santa-Clara 2003b) by defining U.S. recession periods based on the NBER recession data.

 $^{^{26}}$ http://www.nber.org/cycles/recessions.html

²⁷We would like to pronounce at this point that this trade-off is ex post in nature since the "cost" of lower returns has not been considered in the optimization objective, which only aims at the reduction of risk, irrespective of the associated return.

less CMVPs with a higher standard deviation than the value weighted benchmark portfolio. It suggests that the higher return of the CMVPs overcompensates the increased standard deviation. In line with this, all CMVPs performed worse in terms of the aforementioned performance metrics compared to the equally weighted benchmark portfolio, which achieved a higher return at a higher standard deviation than (almost) every CMVP. A completely different impression arises if one considers the factor model based alpha measures. Irrespective of the employed factor model and benchmark portfolio, all factor models achieve a (in some cases substantially) positive alpha. Though this may be encouraging at first sight, it is important to highlight the associated regression statistics as well as the behavior of the alpha measures with respect to the maximum portfolio weight constraint.

Irrespective of the considered factor model, the alpha measure follows a hump shaped pattern with respect to the maximum portfolio weight constraint, which is attributable to the annualized mean return behaivor.²⁸ Though this comovement may seem surprising at first sight, the effect becomes clearer by taking the fairly constant absolute values of systematic risk, as measured by the beta factors, into account. The increasing share of non-systematic risk for less CMVPs, which is at least partially accompanied by increasing returns (up to a maximum portfolio weight of 11%) casts nevertheless doubts upon the postulation that systematic risk is the only source of risk which is rewarded at the market.²⁹

Concluding the empirical results, the CMVPs seem to outperform the value weighted benchmark portfolio based on all considered performance metrics that incorporate a total measure of risk. The opposite of this finding holds for the equally weighted benchmark portfolio. If one assesses the performance based on the alpha measure, all CMVPs seem to outperform both benchmark portfolios. Within the group of CMVPs, the empirical trade-off between risk and return seems to be optimal for portfolios with a maximum portfolio weight in the range between 8% - 11%.

²⁸This is mirrored in a cross correlation of 0.934 (0.935) between the four factor alpha and the equally (value) weighted benchmark portfolio. The cross correlations between the other factor model based alphas and the annualized mean returns are likewise well above 0.9.

²⁹In the ongoing discussion whether idiosyncratic risk has explanatory power concerning the cross section of returns (and may thus be rewarded), Malkiel and Xu (1997; 2004) and Goyal and Santa-Clara (2003b) provide evidence in favor of the paper, while Bali et al. (2005) contradict those results.

4.2 Robust performance metric inference

Inference concerning the thus far derived empirical findings of the CMVP performance is drawn by the bootstrap approaches described in section 2.3. Given the empirical evidence of the non-normality of the CMVP returns, provided by the Jarque-Berra test p-values in table 3, we stress once more the necessity of bootstrap inference. Accordingly any inference for the Sharpe ratio based on the parametric Sharpe ratio test by Jobson and Korkie (1981a) in its corrected version by Memmel (2003) would be incorrect - at least for our data. We turn in the following to the CMVP performance inference relative to the value weighted benchmark portfolio, which is followed by the assessment relative to the equally weighted benchmark portfolio. Beside the bootstrap inference for the annualized mean return and standard deviation of the CMVPs. This is especially important for the assessment whether the empirically observable overcompensation of risk by return, reflected in the empirical performance metric pattern, is statistically significant.³⁰

The bootstrapped p-values in table 4 show that a significantly higher return compared to the value weighted benchmark portfolio, at the 5% (10%) significance level, has only been delivered by the CMVPs with a maximum portfolio weight restriction of 9% - 11% (6% - 14% and 20%). This proves that the empirical observation of higher returns for every CMVP in comparison to the value weighted benchmark portfolio is statistically not significant for most portfolios. Broadly in line with the empirical evidence is the reduction of the standard deviation. The standard deviations of the CMVP with a maximum portfolio weigh constraint of 6% and less are accordingly significantly reduced in comparison to the value weighted benchmark portfolio.

Most interestingly is the assessment whether the empirically observable domination of the return over the risk effect is of statistical significance. Starting with the alpha measures³¹, the empirically observable dominance of the return effect seems to be underpinned by the bootstrap inference. All minimum-variance portfolios in the maximum portfolio weight

³⁰The p-values for the differences in annualized mean returns and standard deviations are derived by the in section 2.3.1 described observation bootstrap for the one sided hypotheses that the difference between the annualized mean return (standard deviation) of the CMVP and the benchmark portfolio (either equally or value weighted) is equal or less than zero:

 $H_0 = \mu_{min-var} - \mu_{benchmark} \le 0$

 $H_0 = \sigma_{min-var} - \sigma_{benchmark} \le 0$

³¹We base in the following all alpha measure descriptions and interpretations on the four factor model based alpha measure, since the four factor model yields the highest adjusted R^2 for all CMVP.

constraint range between 7%-17% (9%-10%) as well as the 20% maximum portfolio weight constraint minimum-variance portfolio yield significant alphas on the 5% (10%) confidence level. Nevertheless, it is important to be aware of the particularities associated with the alpha measure in the context of the minimum-variance portfolio performance mentioned in section 4.1. Taking a closer look, it shows that almost only those portfolios with a statistically significant increase in the annualized mean return achieved a significant alpha. Following this, the significance of the alpha measure for portfolios that have a significant increase in annualized mean return does not come at a surprise, since the systematic risk measure does not mirror the steady increase in total risk.

Based on the empirical results, the empirical Sharpe ratio and CEQ point estimates of every CMVP have been higher for every CMVP. Nevertheless, the bootstrap inference shows that only those portfolios with a maximum portfolio weight constraint of less than 13% (8%) for the Sharpe ratio and 12% (8%) for the CEQ delivered significantly better performance metrics on the 10% (5%) confidence level. Accordingly, the bootstrap reveals that CMVPs with a significantly reduced standard deviation (in comparison to the value weighted benchmark portfolio) deliver significantly higher risk adjusted performance metrics than the benchmark portfolio. Following this, the empirically observable domination of the return over the risk effect may not be considered to be statistically significant. Additional evidence for this is provided by the Sortino ratio. All considered minimum-variance portfolios with a maximum portfolio weight of less than 12% (14%) deliver significantly higher Sortino ratios relative to the value weighted benchmark portfolio on a 5% (10%) confidence level.

The statistical inference of the CMVP performance in comparison to the equally weighted benchmark portfolio reveals almost no surprising insights. The empirically lower performance metrics of the CMVP in comparison to the equally weighted benchmark portfolio are underpinned by the bootstrap results in table 4. Accordingly, the one sided hypothesis that the CEQ, Sharpe and Sortino ratios of the equally weighted benchmark portfolio exceed those of the CMVPs may not be rejected.

A somewhat different impression arises again from the alpha measures, which clearly points at an outperformance of the CMVPs. The problem with the alpha measure based on the equally weighted benchmark portfolio is however the same as already described.

Concluding these first results, the broad picture shows that deterministic diversification,

which is achieved by restrictive maximum portfolio weight constraints, does significantly reduce the realized out-of-sample portfolio risk, as measured by the standard deviation. Nevertheless, it is empirically observable that the imposed maximum portfolio weight constraints lead for the least and most restrictively CMVPs to a decline in the annualized mean return. This is in turn reflected in the empirically declining values of risk adjusted portfolio performance metrics that incorporate a total measure of risk.

The bootstrap inference proves however, that this empirical finding is not significant for the most restrictively CMVPs. Contrary, it turns out that those portfolios with the highest deterministic diversification have the statistically most significant outperformance based on total risk incorporating performance metrics in comparison to the value weighted benchmark portfolio. A somewhat contrarian picture is presented by the systematic risk adjusted alpha measure. Due to the fairly stable share of systematic risk across all CMVPs, those portfolios with a high annualized mean return achieve a statistically significant alpha, irrespective of the considered market proxy (benchmark portfolio). Accordingly one may constitute that certain CMVPs (roughly all minimum-variance portfolios with a portfolio weight constraint of 7% or less) significantly outperform the value weighted benchmark portfolio based on risk adjusted performance.

This observation may not be confirmed in comparison to the the equally weighted benchmark portfolio. Though the most restrictively CMVPs significantly reduce the standard deviation in comparison to the equally weighted benchmark portfolio, none of the CMVPs achieves higher empirical performance metrics. This empirical finding is clearly underpinned by the bootstrap results.

5 Robustness checks

In this section we provide evidence of the constrained minimum-variance investment strategy performance in different market phases. Additionally, we check upon the sensitivity of our results with respect to the portfolio revision frequency. The section is accordingly divided into two parts, whereby we assess the performance of the constrained minimumvariance strategy over three subperiods first. The subperiods comprise in particular U.S. recession periods as well as high volatility and low volatility periods. This is followed by an assessment of the variability of our results with respect to the portfolio revision frequency, which is in the second part of this section varied from three to six and twelve months respectively.

5.1 Subsamples

The assessment of subperiods shall foremost yield insights concerning the robustness of the derived results for the complete sample period and reveal whether the constrained minimum-variance performance over the complete sample period is driven by any specific subperiod. Attention is specifically paid to the best performing CMVP in each subperiod in order to draw conclusions concerning the constancy of the optimal³² portfolio weight constraint. Our choice of subsample periods, namely U.S. recession periods as well as high and low idiosyncratic volatility periods, bases on the work by Kosowski (2004) and Campbell et al. (2001) respectively.

Empirically recession periods are characterized by increased volatility and low (negative) annualized mean returns for the equally (value weighted) benchmark portfolio. Moreover, Kosowski (2004) argues that investors care especially about portfolio performance in recession periods due to the high marginal utility of wealth of investors over these periods. Accordingly improvements in the (risk adjusted) performance relative to the equally and value weighted benchmark portfolios seem to be particularly valuable for investors.

The segregation into high and low idiosyncratic volatility periods follows Campbell et al. (2001) who find a deterministic trend in idiosyncratic firm level volatility we choose to assess the CMVP performance in both volatility market states, with the low idiosyncratic volatility subsample from April 1968 to December 1985 and the high idiosyncratic volatility subsample from January 1986 to December 1997.³³ Campbell et al. (2001) note that the "correlation among individual stock returns declined", while idiosyncratic risk increased over their sample period. Accordingly, possible sensitivities of the CMVP performance to the development of idiosyncratic risk and return correlations are assessed over these two subperiods.

Starting with the U.S. recession periods, the big picture shows clearly the profitability of the constrained minimum-variance approach, resulting in overall higher performance metrics in comparison to both benchmark portfolios. Empirical evidence concerning the

³²Optimal has in this context to be understood in the sense of empirically best performing.

³³Campbell et al. (2001) define the complete low idiosyncratic volatility period from July 1962 to December 1985. Since our sample starts in April 1968, statements concerning the low idiosyncratic volatility period are accordingly based on the period from April 1968 to December 1985.

negative CEQ reveals that more risk averse investors still shy, despite the empirically favorable risk return profile of the CMVPs, investments in any of the portfolios during recession periods. Despite the overall encouraging empirical results, only few CMVPs deliver a statistically significant better portfolio performance relative to the benchmark portfolios. In comparison to the value weighted benchmark portfolio only those CMVPs with a significantly higher return than the benchmark portfolio achieved significantly higher Sharpe and Sortino ratios.³⁴ Accordingly all CMVPs with a maximum portfolio weight constraint of 5% (9%) and less achieved a higher Sharpe and Sortino ratios than the value weighted benchmark portfolio on a 5% (10%) significance level. The favorable results for the more restrictively CMVP are underpined by the alpha measure. Significant alphas are only generated by the CMVPs with a maximum portfolio weight constraint of 4% and less on a 10% confidence level. Similar results for the alpha measure are obtained if one considers the equally weighted benchmark portfolio as market proxy. Contrary, inference concerning the Sharpe and Sortino ratios reveals that the empirically observable outperformance of the equally weighted benchmark portfolio by the CMVPs is in no case statistically significant.

To sum up the findings of this first subperiod assessment, the more restrictively CMVPs achieve significantly higher returns at a lower risk, which is finally reflected in significantly higher performance metrics in comparison to the value weighted benchmark. In comparison to the equally weighted benchmark portfolio, almost every CMVP achieved a substantial risk reduction and an empirically higher annualized mean return. Nevertheless, this does not result in statistically higher performance metrics that base on a total measure of risk. Opposed to that, the most restrictively CMVPs deliver a substantial and significant alpha. All in all, the more restrictively CMVPs seem to perform reasonably well during recession periods.

Turning to the high and low volatility periods³⁵, as defined by Campbell et al. (2001), the empirical characteristics may be confusing at first sight. Both benchmark portfolios exhibit higher standard deviations in the low than in the high volatility period. This effect is attributable to the development of the correlations among stocks. As pointed out by Campbell et al. (2001), the average correlation among stocks declined over time. The

³⁴We do not focus on the CEQ since risk averse investors would, according to the empirically negative CEQ measure, not have valued any of the CMVPs.

³⁵We stress once more that the segregation bases upon the firm level specific and not on the market level volatility.

diversification effect of both benchmark portfolios in the low volatility period, from April 1968 to December 1985, has correspondingly been low, resulting in the comparably high standard deviation of both benchmark portfolios. Turning to the CMVP performance, the value weighted benchmark portfolio delivered in the low volatility period higher performance metrics that incorporate a total measure of risk as well as positive alphas in comparison to the value weighted benchmark portfolio. This observation may only partially be confirmed for the high volatility period, namely for the most restrictively CMVPs. Compared with the equally weighted benchmark portfolio, none of the CMVPs deliver higher performance metrics based on a total measure of risk in the low volatility period. Nevertheless, all CMVPs achieve positive alphas, which holds as well for the high volatility period. Beside the positive alphas, the most restrictively CMVPs achieve additionally Sharpe ratios in excess of the equally weighted benchmark portfolio, while almost every CMVP exhibits higher Sortino ratios than the equally weighted benchmark portfolio.

The bootstrap results in tables 6 and 7 underpin the empirical findings for the high and low volatility periods. The significantly higher returns of the CMVP in comparison to the value weighted benchmark portfolio lead in turn to significantly higher CEQ, Sharpe and Sortino ratios. Even more striking is the high statistical significance of the alpha measures for every CMVP, irrespective of the considered market proxy. Despite the significantly reduced standard deviations of the CMVPs in comparison to the equally weighted benchmark portfolio during the low volatility period, the bootstrap results point at statistically indistinguishable CEQ, Sharpe and Sortino ratios. This shows that the empirically higher performance metrics of the equally weighted benchmark portfolio are not significantly higher than those of the CMVPs.

During the high volatility period, the comparison to both benchmarks shows the benefit of deterministic diversification. The empirical observation of higher performance metrics for the most restrictively CMVPs during this period is clearly underpinned by the bootstrap results. Significantly higher alpha measures on the 5% (10%) confidence level are achieved for the minimum-variance portfolios with a maximum portfolio weight constraint of 2% - 3% (4%), irrespective of the market proxy. Significantly higher Sharpe and Sortino ratios in comparison to both benchmarks may as well be reported for the most restrictively CMVPs. The CMVPs with a maximum portfolio weight constraint of 2% - 3% (2%) achieved significantly higher Sharpe and Sortino ratios in comparison to the equally (value) weighted benchmark portfolio on a significance niveau well below the

10% level.

Concluding the findings from the considered subsamples the results show that there is not the optimal maximum portfolio weight constraint for all subsamples. The results from the subperiods rather convey the impression that the degree of deterministic diversification depends on the level of idiosyncratic risk. Both, the U.S. recession periods as well as the high volatility period are characterized by increased idiosyncratic volatility. Though the average correlation among stocks in these periods is likely to be contrary³⁶, the results show clearly that statistically significant results in both periods may only be derived for more restrictively CMVPs. This is in line with the finding by Campbell et al. (2001), who note that an increased number of stocks is necessary to diversify the high level of idiosyncratic risk.

Though the most restrictively CMVPs were not always the best performing ones, one may nevertheless conclude that the most restrictively CMVPs achieved in every subsample a statistically significant risk reduction in comparison to the equally weighted benchmark portfolio. This shows clearly that the deterministic diversification through maximum portfolio weight constraints additionally reduces the portfolio risk as measured by the standard deviation. Additionally, the result point out that there is not *the* "optimal" maximum portfolio weight constraint. Moreover, the degree of deterministic diversification seems to depend on the respective market phase.

Most results for the considered subperiods are broadly in line with the overall picture. The outperformance of the value weighted benchmark portfolio by the more restrictively CMVPs is especially in U.S. recession and high volatility period observable, while almost every CMVP outperformed the value weighted benchmark portfolio during the low volatility period. The worse performance of the CMVPs in terms of total risk incorporating performance metrics compared to the equally weighted benchmark portfolio is confirmed for almost every subperiod. An exception build the most restrictively CMVPs in the high volatility period, which offer significantly higher performance metrics. Accordingly, we do not find evidence, that the overall findings over the complete sample period depend on a specific subperiod.

³⁶The average correlation among individual stocks in recession periods is usually tight, while the average correlation among individual stocks during the high volatility period is low.

5.2 Portfolio revision frequency

In order to check upon the sensitivity of our results to the portfolio revision frequency for both, the overall sample period as well as the considered subperiods, we vary in the following the revision frequency from three to six and twelve months. Assessing the results for the complete sample period first we find that the standard deviations, as depicted in table 8 show a parallel behavior of the CMVP with different revision frequencies. Remarkable is the relation between the standard deviation and the revision frequency for CMVPs with a maximum portfolio weight constraint of 6% and more. The differences in standard deviations become for those CMVP increasingly large, whereby the CMVP with the highest (lowest) revision frequency exhibit the highest (lowest) standard deviation. This observation is reflected in the empirical performance metrics which nevertheless follow closely the behavior of the annualized mean returns. Correspondingly, the CMVP with a three (twelve) months portfolio revision frequency exhibit the empirically highest (lowest) performance metric values up to the maximum portfolio weight constraint level of 10%. As already observed for the annualized returns, this observation is inverted for CMVP with a maximum portfolio weight constraint of 14% and more, leaving the CMVP with a revision frequency of 12 (3) months with the highest (lowest) performance metric values.

The inference results for the CMVP with six and twelve months revision frequency over the complete sample period yield noteworthy insights. As observed for the CMVP with a three months portfolio revision frequency, a significant reduction in the portfolio standard deviation in comparison to the value weighted benchmark portfolio is only for the most restrictively CMVPs observable. This finding holds for all CMVPs, irrespective of their revision frequency.

While the significance of the performance metrics was in case of the three months revision frequency mainly driven by the significantly lower standard deviation, one may now observe two sources that drive the significance of the performance metrics: the significantly reduced standard deviation and the significantly higher return. Following this, the most restrictively CMVPs (2%-4% maximum portfolio weight constraint) achieve for the six and twelve months revision frequency significantly higher CEQ, Sharpe and Sortino ratios in comparison to the value weighted benchmark portfolio on a significance level well below the 10% level. This is mainly attributable to the significantly reduced standard (and downside-) deviation. The effect of the significantly increased return is in turn reflected in the significant performance metrics for the less constrained portfolios. Accordingly, all CMVP with a six months portfolio revision frequency and maximum portfolio weight constraints between 7% - 13% deliver significantly³⁷ higher CEQ, Sharpe and Sortino ratios in comparison to the value weighted benchmark portfolio. The results for the CMVP with a twelve months portfolio revision frequency point in the same direction and deliver even more pronounced findings concerning the driving forces of the performance metric significance.

The factor model based alpha measures are over the complete sample period invariant to both, the market proxy and the portfolio revision frequency. Accordingly, the in section 4.1 reported alpha pattern³⁸ emerges for all settings, which results in the reduced importance of the alpha measure for the CMVP assessment in the complete sample period due to the in section 4.1 elaborated shortcomings associated with this pattern. Further invariance to the portfolio revision frequency is observable for the performance metrics in comparison to the equally weighted benchmark portfolio. Though all CMVP deliver significantly lower standard deviations than the equally weighted benchmark portfolio, the hypothesis of higher performance metrics for the equally weighted benchmark portfolio may in no case be rejected.

Concluding the results for the complete sample period, the qualitative findings from section 4.2 are confirmed. The constrained minimum-variance approach delivers, irrespective of the portfolio revision frequency, higher risk adjusted performance than the value weighted benchmark portfolio. Broad evidence for this finding is provided by the reported CEQ, Sharpe and Sortino ratios. The driving source for the significantly higher CEQ, Sharpe and Sortino ratios varies however with the portfolio revision frequency. The significance of the higher performance metrics of the CMVPs with a three months portfolio revision frequency in comparison to the value weighted benchmark portfolio is solely driven by the significantly reduced standard deviation. This changes for the CMVP with portfolio revision frequencies of six and twelve months. Though the standard deviation of the CMVPs is in both cases significantly reduced in comparison to the value weighted benchmark portfolio, the significance of the performance metrics is mainly driven by the significantly higher annualized return in comparison to the value weighted benchmark

 $^{^{37}\}mathrm{Mostly}$ below the 5% but always well below 10% confidence level.

³⁸Significant alphas are only achieved by less CMVPs which bare a large share of idiosyncratic risk

portfolio.³⁹ The reason for the significantly higher risk reduction observed for the twelve months portfolio revision frequency remains open. The explanation for the lower standard deviation of the twelve months revision frequency based CMVPs in comparison to the higher revision frequencies based CMVPs must be founded in better out-of-sample properties of variance-covariance estimate, which is estimated accoring to the portfolio revision frequency. Following this, it is suggestive that the cummulative estimation error of four quarterly variance-covariance estimates exceeds that of a single estimate for an annual period.

The results for the U.S. recession periods⁴⁰ underpin the high influence of the revision frequency on the portfolio performance. The variation of the initial portfolio revision frequency from three to six months leads to significantly higher returns for the less restrictively CMVPs. In turn, the initial outperformance of the value weighted benchmark portfolio by the CMVPs with a revision frequency of three months and maximum portfolio weight constraints of less than 10% is extended by the CMVPs with a six months portfolio revision frequency. Accordingly, the CMVPs with maximum portfolio weight constraints between 12% - 19% achieve significantly higher CEQ, Sharpe and Sortino ratios than the value weighted benchmark portfolio, which are also according to amount higher than those of the CMVPs with a three months portfolio revision frequency. The further variation of the portfolio revision frequency to twelve months changes the results dramatically. Though the most restrictively CMVPs still reduce the standard deviation substantially, none of the considered performance metrics exceeds the corresponding value weighted benchmark portfolio significantly. Compared to the equally weighted benchmark portfolio, almost⁴¹ none of the CMVPs delivers significantly higher performance metrics. Broadly invariant results with respect to the portfolio revision frequency are obtained for the low volatility periods, as defined by Campbell et al. (2001). Irrespective of the portfolio revision frequency and maximum portfolio weight constraint, almost every CMVP outperforms the value weighted benchmark portfolio based on every considered performance metric in the low volatility period. Additionally, all CMVP deliver significant

³⁹This may be deducted from the bootstrap inference in tables 9 and 10, which show that almost all significantly higher performance metrics on the 5% level are associated with significantly higher returns, whereas the reduction of the standard deviation is not significant.

⁴⁰The tables for the considered subperiods with different portfolio revision frequencies are not provided but are available from the authors upon request.

⁴¹Only limited and very weak statistical evidence of an outperformance of the equally weighted benchmark portfolio by the less CMVPs with a revision frequency of 6 months is provided by the Sortino ratio.

alphas, irrespective of the portfolio revision frequency and market proxy, while none of the CMVP outperforms the equally weighted benchmark portfolio.

Somewhat mixed results are obtained for the high volatility period. The empirical performance metric values show that the most favorable portfolio weight constraints for the CMVPs with three and six months revision frequencies, are the most restrictive ones. Contrary, the CMVPs with a twelve months revision frequency achieve the largest performance metric values for the less restrictively portfolio weight constraints. Given these empirical differences, the bootstrap reveals that the most restrictively CMVP with the three months portfolio revision frequency achieves the most significant and according to amount highest risk adjusted performance based on any of the considered performance metrics.

Evidence from the three subsamples shows that not only the choice of the maximum portfolio weight constraint but also the portfolio revision frequency has substantial influence on the portfolio performance. Moreover, it shows that the best performing revision frequency for the complete sample period is not optimal for every considered subperiod. According to our afore defined criterion for assessing the "optimal" portfolio revision frequency, we find that the six months portfolio revision frequency is the best performing. During the high (low) volatility period a portfolio revision frequency of three and six (six and twelve) months would have been optimal.

6 Conclusion

This paper analyzed the risk-adjusted performance of constrained minimum-variance portfolios in the U.S. during the period from April 1968 to December 2007. We employed the constrained minimum-variance approach by Jagannathan and Ma (2003) on a rolling sample basis and vary the maximum portfolio weight constraint in one percentage steps between 2% - 20%. The implementation of maximum portfolio weight constraints assures thereby a certain degree of diversification and achieves a shrinkage like effect. We assess the CMVP performance by several performance metrics in order to provide a comprehensive picture. In particular, we provide evidence of the constrained minimum-variance portfolio performance on the certainty equivalent, the Sharpe and Sortino ratio as well as on factor model based alpha measures. We test upon the significance of these performance metrics by employing a nonparametric bootstrap. Bootstrap methods are required for robust inference concerning the performance metrics since the CMVP returns are not normally distributed.

Since earlier papers either focused on the descriptive performance metric comparison (e.g. Bloomfield et al. 1977 and Chan et al. 1999) or employed invalid inference methodologies (DeMiguel et al. (2007), a statistically robust assessment concerning the profitability of the (constrained) minimum-variance approach has not been done. Additionally, none of the prior studies based the portfolio optimization on a complete U.S. equity universe. Our paper closes thereby the lack of robust inference concerning the profitability of the (constrained) minimum-variance approach.

Our empirical results over the complete sample period show that the imposed maximum portfolio weight constraints reduce the out-of-sample standard deviation. The resulting performance metric values of the the more restrictively CMVP exceed those of the value weighted benchmark portfolio significantly, while they are well below those of the equally weighted benchmark. The assessment of various subperiods, namely the U.S. recession periods, as well as low and high volatility periods broadly confirmed the significant outperformance of the value weighted benchmark portfolio by the more restrictively CMVPs. The broad result of the significant outperformance of the value weighted benchmark portfolio by the CMVPs holds even for different portfolio revision frequencies, whereas the lowest revision frequency yields surprisingly the most favorable results. Sobering results are obtained in comparison to the equally weighted benchmark. The CMVP achieved in no setting noteworthy statistically higher performance metrics in comparison to the equally weighted portfolio.

Nevertheless, the high sensitivity of the CMVPs to the portfolio revision frequency remains thus far unexplored. Highlighted should in this context the obvious link between the portfolio revision frequency and the behavior of the standard deviations. Last but not least remains the question for *the* optimal maximum portfolio weight constraint. Although the results of Bloomfield et al. (1977) and Campbell et al. (2001) suggest that a reasonably diversified portfolio should consist of at least 25 or - nowadays that stock markets became more volatile - 50 stocks, it seems that these guidelines are not a panacea for the choice of the "optimal" constraint for minimum-variance investing.

The table report April 1968 to French's websi heteroscedasti	orts the Decemb- ite: http city usin	regressic er 2007. ://mba. ig the au	n statis The mé .tuck.d	tics for 1 arket fac artmout ation an	the consition $\mathbf{h} \in \mathbf{h}$ to $\mathbf{h} \in \mathbf{d} \mathbf{u} / \mathbf{f}$	idered of sreby pro sedes/fs	ne, three oxied by aculty/l city cons	and fou the CR3 xen.fre sistent c	ir factor SP value inch/dat ovarianc	models e weighte ca_libre e matrix	(as desc ed mark ary.htm t from N	rribed ir et index 1. The lewey an	i section . All fa provided nd West	(2.3.2) ctor time to the transformed to the transformed (1987).	over the le series es are ac	comple data ar ljusted f	te samp e taken or autoo	le perioo from Ke correlati	d from enneth on and
									Maxim	um Weigh	its								
	2%	3%	4%	5%	6%	7%	8%	9%	10%	11%	12%	13%	14%	15%	16%	17%	18%	19%	20%
<u>1 Factor Model</u> Market	1 706	0.667	0 655 0	0.651	0.650	0.643	0.616	0.634	0.656	0.635	0.632	0 501	0.649	0.643	0.621	0 622	0 597	0.601	0 669
t-Value	16.873	15.209	14.202	12.979	12.220	10.834	10.043	9.841	9.902	9.080	8.904	8.428	9.065	8.788	8.495	8.247	7.959	7.135	7.607
Adj. R^2	0.701	0.641	0.579	0.539	0.508	0.459	0.408	0.393	0.388	0.350	0.337	0.288	0.326	0.317	0.293	0.285	0.254	0.240	0.272
3 Factor Model																			
Market	- 0.768	0.724	0.702	0.694	0.692	0.686	0.651	0.668	0.688	0.663	0.665	0.615	0.675	0.673	0.644	0.642	0.637	0.618	0.696
t-Value	19.921	17.606	16.238	14.329	13.699	11.945	10.938	10.668	10.839	9.950	9.659	8.892	9.906	9.604	9.140	8.805	8.538	7.473	8.080
SMB	0.033	0.018	0.031	0.044	0.025	0.026	0.010	0.021	-0.004	-0.009	-0.032	0.025	-0.053	-0.058	-0.041	-0.027	-0.028	-0.003	-0.028
t-Value	0.841	0.419	0.681	0.878	0.478	0.444	0.165	0.333	-0.060	-0.130	-0.496	0.387	-0.810	-0.877	-0.640	-0.413	-0.418	-0.044	-0.350
HML	0.241	0.213	0.187	0.182	0.165	0.169	0.128	0.137	0.108	0.094	0.093	0.104	0.074	0.062	0.051	0.049	0.117	0.056	0.075
t-Value	3.680	3.113	2.488	2.284	1.928	1.809	1.335	1.303	0.950	0.804	0.816	0.902	0.606	0.490	0.400	0.370	0.852	0.382	0.479
Adj. R^2	0.728	0.662	0.594	0.552	0.517	0.468	0.412	0.397	0.389	0.350	0.337	0.288	0.326	0.317	0.292	0.283	0.255	0.238	0.271
4 Factor Model																			
Market	- 0.777	0.733	0.712	0.708	0.706	0.701	0.668	0.685	0.706	0.681	0.685	0.627	0.693	0.692	0.662	0.659	0.653	0.632	0.712
t-Value	20.939	18.191	16.834	15.037	14.361	12.569	11.562	11.217	11.376	10.432	10.157	9.093	10.330	9.965	9.428	9.010	8.610	7.514	8.129
SMB	0.034	0.019	0.032	0.045	0.026	0.027	0.011	0.022	-0.002	-0.007	-0.031	0.026	-0.051	-0.057	-0.039	-0.025	-0.027	-0.002	-0.026
t-Value	0.813	0.415	0.692	0.891	0.496	0.474	0.193	0.356	-0.037	-0.109	-0.485	0.400	-0.787	-0.850	-0.607	-0.379	-0.392	-0.028	-0.321
HML	0.256	0.227	0.205	0.206	0.190	0.195	0.158	0.166	0.140	0.125	0.127	0.124	0.106	0.094	0.083	0.080	0.145	0.080	0.102
t-Value	3.728	3.154	2.626	2.550	2.165	2.054	1.643	1.561	1.207	1.058	1.112	1.027	0.843	0.719	0.620	0.572	1.014	0.512	0.619
MOM	0.063	0.061	0.075	0.104	0.104	0.113	0.126	0.124	0.133	0.132	0.145	0.086	0.134	0.136	0.133	0.129	0.121	0.102	0.118
t-Value	1.394	1.297	1.536	2.032	1.769	1.807	1.950	1.828	1.862	1.822	2.106	1.107	1.743	1.682	1.582	1.459	1.356	1.028	1.138
Adj. R^2	0.732	0.666	0.599	0.563	0.527	0.478	0.425	0.408	0.401	0.361	0.350	0.291	0.336	0.327	0.302	0.292	0.262	0.242	0.276

Table 1: Regression statistics for the constrained minimum-variance portfolios with 3 months portfolio revision frequency for the period from April 1968 to December 2007 with a value weighted market proxy

26

April 1968 to French's websi heteroscedasti	orts the Decemb ite: htt city usir	regressioner 2007. P://mba ng the au	The ma The ma .tuck.d .tocorrel	arket fac arket fac lartmout lation ar	une cons tor is he zh.edu/ id heter	sucerea c reby pro pages/f oscedast	ne, unre oxied by aculty/ icity con	e and ro the CR5 'ken.fr6 sistent c	ur tacto. SP equa. ench/da ovarian.	r models lly weigh .ta_libr ce matri	ted mar ted mar ary.hti x from I	crubeu L 'ket inde nl. The Newey a	a secuor sx. All f provide nd West	actor tir actor tir d t-value (1987).	over une ne series es are ac	s data al ljusted j	re taken cor autoo	le perio from K correlati	a rrom enneth on and
									Maxin	num Weig	hts								
	2%	3%	4%	5%	6%	7%	8%	9%	10%	11%	12%	13%	14%	15%	16%	17%	18%	19%	20%
1 Factor Model																			
Market	0.465	0.435	0.429 10.121	0.428	0.426	0.419 0 656	0.401	0.416 0.001	0.426	0.408 7 07E	0.403 7 660	0.394 7 500	0.403 $\gamma \ e^{\gamma E}$	0.400	0.390 7 001	0.394	$0.377 e^{-200}$	0.389 6 E 01	0.424 6 560
V^{-Vulue} Adj. R^2	0.495	0.444	0.404	9.000	0.354	0.317	0.280	0.276	0.266	0.235	0.222	0.208	0.208	0.199	0.187	0.185	0.163	0.163	0.177
3 Factor Model																			
Market	0.652	0.613	0.591	0.583	0.587	0.577	0.553	0.571	0.591	0.565	0.571	0.535	0.576	0.572	0.546	0.545	0.537	0.528	0.590
t-Value	12.363	11.568	10.692	9.738	9.474	8.501	8.109	8.186	8.359	7.812	7.751	7.546	7.924	7.749	7.434	7.255	7.082	6.437	6.655
SMB	-0.492	-0.476	-0.444	-0.424	-0.447	-0.438	-0.436	-0.440	-0.482	-0.465	-0.494	-0.410	-0.518	-0.519	-0.481	-0.466	-0.460	-0.430	-0.503
t-Value	-6.288	-5.998	-5.013	-4.435	-4.553	-3.982	-3.912	-3.898	-4.143	-3.814	-4.066	-3.639	-4.287	-4.260	-4.128	-3.996	-4.005	-3.427	-3.872
HML	-0.012	-0.026	-0.045	-0.048	-0.063	-0.058	-0.085	-0.081	-0.116	-0.124	-0.123	-0.094	-0.147	-0.159	-0.160	-0.162	-0.094	-0.146	-0.154
t-Value	-0.152	-0.322	-0.515	-0.530	-0.661	-0.569	-0.809	-0.720	-0.948	-0.985	-1.012	-0.789	-1.124	-1.171	-1.195	-1.180	-0.680	-0.981	-0.971
Adj. R^2	0.591	0.535	0.479	0.443	0.421	0.376	0.337	0.328	0.325	0.289	0.281	0.245	0.269	0.260	0.239	0.233	0.204	0.198	0.221
4 Factor Model																			
Market	-0.694	0.652	0.634	0.633	0.637	0.630	0.608	0.627	0.650	0.623	0.633	0.579	0.635	0.631	0.604	0.601	0.590	0.576	0.644
t-Value	17.379	15.677	14.470	13.250	12.907	11.311	10.903	10.678	10.999	10.015	9.968	9.363	10.408	10.104	9.562	9.190	8.767	7.847	8.121
SMB	-0.531	-0.512	-0.484	-0.471	-0.495	-0.487	-0.488	-0.493	-0.538	-0.519	-0.552	-0.451	-0.573	-0.575	-0.535	-0.518	-0.510	-0.475	-0.554
t-Value	- 7.690	-7.215	-6.249	-5.682	-5.861	-5.134	-5.071	-5.087	-5.456	-4.934	-5.419	-4.585	-5.559	-5.472	-5.297	-5.088	-5.203	-4.183	-4.766
HML	0.025	0.009	-0.007	-0.004	-0.018	-0.012	-0.037	-0.033	-0.064	-0.073	-0.069	-0.056	-0.095	-0.107	-0.110	-0.113	-0.047	-0.104	-0.107
t-Value	0.299	0.103	-0.082	-0.041	-0.190	-0.117	-0.348	-0.286	-0.518	-0.580	-0.569	-0.448	-0.713	-0.769	-0.794	-0.786	-0.325	-0.656	-0.634
MOM	0.156	0.148	0.160	0.189	0.190	0.198	0.209	0.210	0.223	0.217	0.233	0.166	0.222	0.222	0.216	0.212	0.201	0.181	0.205
t-Value	2.488	2.358	2.501	2.878	2.626	2.590	2.689	2.594	2.608	2.535	2.856	1.860	2.461	2.372	2.233	2.102	1.988	1.644	1.771
Adj. R^2	0.617	0.558	0.504	0.477	0.453	0.408	0.372	0.360	0.358	0.319	0.314	0.261	0.297	0.288	0.265	0.257	0.225	0.213	0.239

Table 2: Regression statistics for the constrained minimum-variance portfolios with 3 months portfolio revision frequency for the period from April 1968 to December 2007 with an equally weighted market factor proxy The table reports the regression statistics for the considered one, three and four factor models (as described in section 2.3.2) over the complete sample period from

27

The table reports over The table reports su a single stock may h returns are normally models. Hereby dem factor benchmark foi the value weighted n performance compar	une p mmarizi ave in tl distribu otes "Alı te very <i>i</i> narket b ison, rep	period ing portf he portf pha 1 Fa alpha is ported as ported as	folio stat folio stat olio. Th ne certain actor" th reported rk portfi s well.	Apt.II tistics as e summinty equinative e alpha e alpha olio (pro-	U 200 U 2000 U well as ary stati ary stati ary alent m measure kets, wl xied by	U Deco the ann stics shc easure (based o here EW the CR	CLEDEL CLEQ) is we the m we the m CLEQ) is n the on denotes SP values	ZUUI mpirical maintean nun reportec factor s the equ	perform nber of s l for a ri model, v ed marke Maxim	nance me tocks in sk avers while "A ghted m st index um Wei	etrics for the por ion coeff lpha 3 F lpha 3 F arket be). The _F	the control the control of the contr	strained well as 1 $\gamma = 4$. <i>l</i> nd "Alp c portfol nce methor	minimu she Jarq Alpha m ha 4 Fac io (prox rics for	m-varia ue-Berra easures a easures do toor" do ied by tl ooth ber	test portf test p-v tre comp so for th ne CRSF ichmark	olios wit alues foi uted for e three a e equally portfolic	h the r t the hy the in s the in s nd four weight s are, f	espective m pothesis th section 2.2. factor moc ed market for the purp	aximum at the pc lescribed lel. The j andex) an ose of vj Benchr	weight actor factor narket kd VW s-a-vis s-a-vis aarks
	2%	3%	4%	5%	6%	7%	8%	6%	10%	11%	12%	13%	14%	15%	16%	7% 1	8% 19	9% 2	0%	νw	EW
Mean no. of Stocks	53	38	29	25	22	20	18	17	16	16	15	14	14	14	13	13	12	5	12		
Jarque-Berra p-values	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000 (000.0	0.000	.000 0.	000 000	0 000	.000	0.000	0.000
Mean	0.071	0.072	0.078	0.078	0.083	0.086	0.090	0.093	0.097	0.097	0.094	060.0	0.091 (.088 (0.086 C	.089 0.	0 270	0 0	.104	0.056	0.144
Std. Dev.	0.132	0.130	0.134	0.138	0.142	0.148	0.150	0.157	0.164	0.167	0.169	0.171	0.175 (.177 (0.178 0	.181 0.	184 0.	190 0	.199	0.156	0.198
CEQ	0.037	0.039	0.042	0.040	0.042	0.043	0.045	0.044	0.043	0.041	0.037	0.031	0.030 (0.025 (0.23 C	.024 0.	010 0.0	010 0	.025	0.008	0.065
Sharpe Ratio	0.541	0.558	0.581	0.566	0.583	0.584	0.600	0.592	0.592	0.582	0.557	0.525	0.519 (.497 (.486 0	.493 0.	421 0.	431 0	.523	0.362	0.725
Sortino Ratio	0.955	0.998	1.063	1.023	1.050	1.062	1.098	1.110	1.105	1.099	1.049	0.998	0.971 (.926 (0.890 0	.906 0.	786 0.3	801 1	.003	0.514	1.056
Alpha 1 Factor (EW)	0.031	0.035	0.041	0.041	0.046	0.050	0.056	0.057	0.061	0.062	0.060	0.056	0.056 (0.054 (0.053 0	.055 0.	045 0.	049 0	.068		
Alpha 1 Factor (VW)	0.031	0.035	0.041	0.041	0.046	0.050	0.055	0.057	0.060	0.061	0.059	0.056	0.055 (0.052 (0.051 0	.054 0.	044 0.	048 0	.066		
Alpha 3 Factor (EW)	0.026	0.031	0.039	0.039	0.045	0.049	0.056	0.057	0.062	0.065	0.062	0.057	0.060 (0.058 (0.058 0	.061 0.	046 0.	053 0	.072		
Alpha 3 Factor (VW)	0.014	0.020	0.028	0.028	0.034	0.038	0.046	0.048	0.052	0.055	0.052	0.049	0.050 (0.048 (0.048 0	.051 0.	036 0.1	044 0	.061		
Alpha 4 Factor (EW)	0.006	0.012	0.018	0.015	0.021	0.024	0.029	0.031	0.034	0.037	0.032	0.036	0.032 (0:030 (0.031 0	.034 0.	020 0.	030 0	.046		
Alpha 4 Factor (VW)	0.007	0.013	0.019	0.016	0.022	0.025	0.031	0.033	0.037	0.039	0.035	0.039	0.034 (0.032 (0.032 0	.036 0.	021 0.	032 0	.048		

Table 3: Empirical statistics and performance metrics for the constrained minimum-variance portfolios with 3 months portfolio revision from the portfolio from April 1068 to December 2007

									Maxir	num Weig.	hts								
	2%	3%	4%	5%	6%	7%	8%	%6	10%	11%	12%	13%	14%	15%	16%	17%	18%	19%	20%
Panel I: Ve	alue weighte	d benchma	٠rk																
Δ Mean	0.152	0.155	0.117	0.125	0.086	0.066	0.058	0.047	0.039	0.046	0.062	0.094	0.085	0.116	0.132	0.116	0.215	0.196	0.051
Δ Std. Dev.	1.000	1.000	1.000	1.000	0.997	0.938	0.837	0.461	0.150	0.104	0.066	0.046	0.020	0.008	0.008	0.004	0.003	0.001	0.001
ΔCEQ	0.026	0.027	0.028	0.044	0.038	0.046	0.047	0.052	0.055	0.076	0.109	0.188	0.182	0.230	0.270	0.267	0.473	0.465	0.278
Δ Sharpe Ratio	0.032	0.028	0.031	0.057	0.044	0.049	0.051	0.051	0.051	0.065	0.095	0.148	0.139	0.183	0.218	0.200	0.358	0.339	0.160
Δ Sortino Ratio	0.024	0.020	0.017	0.039	0.036	0.037	0.040	0.031	0.035	0.043	0.066	0.093	0.110	0.139	0.180	0.164	0.272	0.270	0.111
Alpha 1 Factor	0.014	0.004	0.005	0.002	0.005	0.000	0.000	0.001	0.002	0.002	0.003	0.007	0.008	0.016	0.017	0.017	0.046	0.047	0.009
Alpha 3 Factor	0.018	0.010	0.005	0.003	0.002	0.000	0.000	0.000	0.000	0.000	0.000	0.001	0.001	0.002	0.003	0.001	0.027	0.023	0.001
Alpha 4 Factor	0.333	0.203	0.118	0.172	0.117	0.085	0.053	0.051	0.049	0.034	0.060	0.052	0.069	0.092	0.098	0.077	0.188	0.120	0.038
Panel II: Eq	ually weigh:	ted benchn	nark																
Δ Mean	0.744	0.737	0.654	0.652	0.568	0.515	0.448	0.395	0.363	0.360	0.406	0.467	0.439	0.468	0.483	0.450	0.596	0.548	0.283
Δ Std. Dev.	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	0.998	0.989	0.976	0.953	0.939	0.909	0.866	0.746	0.510
ΔCEQ	0.092	0.083	0.073	0.096	0.086	0.088	0.080	0.100	0.103	0.120	0.158	0.223	0.223	0.259	0.287	0.289	0.451	0.465	0.284
Δ Sharpe Ratio	0.170	0.159	0.147	0.177	0.139	0.151	0.136	0.154	0.149	0.175	0.217	0.292	0.287	0.323	0.355	0.344	0.517	0.507	0.277
Δ Sortino Ratio	0.187	0.162	0.128	0.162	0.145	0.148	0.131	0.133	0.130	0.140	0.186	0.235	0.251	0.282	0.326	0.307	0.456	0.435	0.240
Alpha 1 Factor	0.008	0.003	0.002	0.003	0.002	0.000	0.001	0.000	0.002	0.001	0.002	0.003	0.009	0.014	0.018	0.016	0.049	0.051	0.010
Alpha 3 Factor	0.020	0.008	0.004	0.005	0.001	0.001	0.000	0.000	0.001	0.000	0.000	0.002	0.001	0.000	0.001	0.000	0.030	0.017	0.001
Alpha 4 Factor	0.344	0.220	0.110	0.169	0.121	0.094	0.055	0.057	0.047	0.027	0.057	0.060	0.071	0.096	0.096	0.080	0 192	0.116	0.033

Alpha, Alpha, Alpha, Alpha, Alpha, Alphark, Alph	0.020 0.285	-0.142 0.071 0.137	0.147 0.219 0.671 0.671 1.100	0.138 0.162 0.085 0.850 0.983
re in the po by denotes ' nchmark fo narket ben netrics for t 1986 to Det <u>Benchn</u>	-0.063 0.219	-0.158 -0.287 -0.521	0.031 0.163 0.189 0.189 0.307 0.307	0.104 0.144 0.062 0.720 0.832
k may hav dels. Herel c factor be c factor be formance 1 n January	$0.112 \\ 0.264$	-0.056 0.313 0.610 0.095 0.066 0.066 0.022 0.019	$\begin{array}{c} 0.086\\ 0.1188\\ 0.115\\ 0.456\\ 0.885\\ 0.040\\ 0.073\\ 0.055\\ 0.054\\ 0.084\\ 0.084\\ 0.084\end{array}$	0.098 0.165 0.165 0.043 0.593 0.593 0.062 0.033 0.033 0.033 0.033 0.034 0.024 0.024
r ther more market is value is per is per is 19%	$0.108 \\ 0.249$	-0.045 0.314 0.609 0.094 0.064 0.064 0.064 0.038 0.037	$\begin{array}{c} 0.063\\ 0.176\\ 0.0176\\ 0.001\\ 0.358\\ 0.655\\ 0.020\\ 0.034\\ 0.034\\ 0.051\\ 0.058\\ 0.056\\ 0.056\\ 0.058\\ 0.056\\ 0.058\\ 0.056\end{array}$	$\begin{array}{c} 0.061\\ 0.160\\ 0.116\\ 0.010\\ 0.382\\ 0.675\\ 0.032\\ 0.012\\ 0.012\\ 0.012\\ 0.014\\ 0.010\end{array}$
tht a simple fauthor a simple fauthor	0.183 0.232	$\begin{array}{c} 0.046\\ 0.661\\ 1.400\\ 0.171\\ 0.173\\ 0.180\\ 0.178\\ 0.178\\ 0.155\\ 0.150\\ 0.150\end{array}$	$\begin{array}{c} 0.074\\ 0.165\\ 0.165\\ 0.448\\ 0.448\\ 0.033\\ 0.033\\ 0.047\\ 0.053\\ 0.047\\ 0.047\\ 0.047\\ 0.040\end{array}$	$\begin{array}{c} 0.033\\ 0.161\\ -0.019\\ 0.206\\ 0.349\\ 0.003\\ -0.017\\ -0.013\\ -0.013\\ -0.013\\ -0.013\end{array}$
um weig 2.2 descr tor mod lex) and iports th ports th ow volat , report	$0.090 \\ 0.231$	-0.046 0.261 0.469 0.074 0.074 0.042 0.041 0.016 0.014	$\begin{array}{c} 0.062\\ 0.167\\ 0.007\\ 0.073\\ 0.373\\ 0.646\\ 0.021\\ 0.034\\ 0.052\\ 0.054\\ 0.054\\ 0.054\end{array}$	$\begin{array}{c} 0.092\\ 0.153\\ 0.153\\ 0.045\\ 0.603\\ 1.071\\ 0.054\\ 0.029\\ 0.026\\ 0.026\\ 0.020\\ 0.020\\ 0.020\\ 0.020\\ 0.020\\ 0.020\\ 0.012\end{array}$
t maxim section (four fac trket inc nel II re for the l nparison	0.090 0.227	-0.043 0.264 0.465 0.074 0.074 0.041 0.040 0.016 0.013	$\begin{array}{c} 0.053\\ 0.165\\ -0.001\\ 0.323\\ 0.548\\ 0.548\\ 0.013\\ 0.013\\ 0.043\\ 0.031\\ 0.031\\ 0.031\\ 0.031\end{array}$	$\begin{array}{c} 0.096\\ 0.154\\ 0.048\\ 0.621\\ 1.104\\ 0.058\\ 0.031\\ 0.031\\ 0.026\\ 0.026\\ 0.022\\ 0.014\end{array}$
spective c the in tree and hted ma iods, pa metrics nree com	0.088 0.232	-0.049 0.251 0.454 0.070 0.072 0.034 0.033 0.015 0.013	$\begin{array}{c} 0.053\\ 0.167\\ -0.004\\ 0.314\\ 0.545\\ 0.545\\ 0.012\\ 0.024\\ 0.045\\ 0.032\\ 0.032\\ 0.035\end{array}$	0.100 0.156 0.051 0.639 1.141 0.639 0.031 0.031 0.037 0.033 0.019
h the re outed fo r the th ally weig ssion per ssion per rmance	0.095 0.227	-0.037 0.290 0.520 0.079 0.080 0.041 0.041 0.018 0.016	$\begin{array}{c} 0.058\\ 0.166\\ 0.166\\ 0.003\\ 0.351\\ 0.610\\ 0.018\\ 0.018\\ 0.037\\ 0.037\\ 0.050\\ 0.051\\ 0.051\\ 0.051\\ 0.041 \end{array}$	$\begin{array}{c} 0.096\\ 0.155\\ 0.048\\ 0.621\\ 1.108\\ 0.055\\ 0.055\\ 0.032\\ 0.019\\ 0.013\\ 0.013\\ 0.013\end{array}$
alios wit are comp do so fo LSP equa .S. reces al perfo is-a-vis]	0.085 0.210	-0.033 0.264 0.481 0.076 0.075 0.044 0.033 0.030	$\begin{array}{c} 0.061\\ 0.165\\ 0.165\\ 0.007\\ 0.370\\ 0.675\\ 0.021\\ 0.034\\ 0.034\\ 0.034\\ 0.034\\ 0.034\\ 0.034\\ 0.034\\ 0.046\\ 0.046\end{array}$	$\begin{array}{c} 0.082\\ 0.147\\ 0.038\\ 0.555\\ 0.982\\ 0.045\\ 0.026\\ 0.026\\ 0.017\\ 0.017\\ 0.017\end{array}$
ce portf aasures a Factor" the CF ics for U ics for U ics for U ics of v ose of v	$0.082 \\ 0.216$	-0.040 0.245 0.450 0.061 0.062 0.024 0.015 0.014	$\begin{array}{c} 0.059\\ 0.163\\ 0.163\\ 0.006\\ 0.636\\ 0.636\\ 0.019\\ 0.019\\ 0.031\\ 0.034\\ 0.034\\ 0.033\\ 0.035\\ 0.035\end{array}$	0.091 0.148 0.048 0.624 0.624 1.107 1.107 0.051 0.025 0.023 0.013 0.013 0.007
n-varian Jpha 4 Jpha 4 Jupha 4 oxied by oxied by the metr the purp	0.080 0.215	-0.042 0.234 0.415 0.066 0.066 0.032 0.031 0.023 0.021	$\begin{array}{c} 0.054\\ 0.160\\ 0.160\\ 0.334\\ 0.575\\ 0.575\\ 0.014\\ 0.026\\ 0.042\\ 0.042\\ 0.034\end{array}$	$\begin{array}{c} 0.101\\ 0.149\\ 0.057\\ 0.678\\ 1.222\\ 0.678\\ 1.222\\ 0.058\\ 0.036\\ 0.036\\ 0.036\\ 0.019\\ 0.010\end{array}$
minimur $\gamma = 4. A$ and "A and "A and "A and "I have a subsection of the second structure st	0.079 0.213	-0.041 0.231 0.418 0.067 0.066 0.032 0.031 0.024 0.022	$\begin{array}{c} 0.063\\ 0.164\\ 0.164\\ 0.009\\ 0.679\\ 0.679\\ 0.022\\ 0.035\\ 0.035\\ 0.035\\ 0.049\\ 0.041\\ 0.041\end{array}$	$\begin{array}{c} 0.089\\ 0.143\\ 0.143\\ 0.048\\ 0.622\\ 1.062\\ 1.062\\ 0.016\\ 0.016\\ 0.016\\ 0.016\\ -0.001\end{array}$
Factor" Factor" 5 benchm inical pe 001), wh nmarks a	0.096 0.211	-0.022 0.316 0.602 0.085 0.084 0.054 0.053 0.053	$\begin{array}{c} 0.065\\ 0.162\\ 0.162\\ 0.013\\ 0.402\\ 0.736\\ 0.024\\ 0.037\\ 0.035\\ 0.035\\ 0.035\\ 0.043\end{array}$	0.075 0.137 0.038 0.551 0.920 0.035 0.011 0.011 -0.014 -0.014
he const an coeffic Alpha 3 Amarket the emp the emp th bencl th bencl	0.090 0.196	-0.016 0.310 0.555 0.079 0.077 0.045 0.044 0.048 0.048	0.056 0.158 0.006 0.353 0.628 0.016 0.016 0.016 0.031 0.031 0.032 0.032	$\begin{array}{c} 0.098\\ 0.130\\ 0.064\\ 0.756\\ 1.285\\ 0.058\\ 0.033\\ 0.033\\ 0.033\\ 0.023\\ 0.018\\ 0.010\end{array}$
t ics for t c aversic while " weightee by Cam by Cam is for bo	0.097 0.205	-0.016 0.331 0.619 0.086 0.086 0.055 0.055 0.055 0.055	$\begin{array}{c} 0.058\\ 0.160\\ 0.007\\ 0.0654\\ 0.654\\ 0.017\\ 0.017\\ 0.030\\ 0.030\\ 0.028\\ 0.028\\ 0.033\\ 0.033\end{array}$	$\begin{array}{c} 0.090\\ 0.133\\ 0.055\\ 0.676\\ 1.131\\ 0.047\\ 0.019\\ 0.019\\ 0.007\\ 0.007\\ -0.003\end{array}$
ice metr or a rish model, equally Panel I defined e metric	0.092 0.200	-0.018 0.312 0.575 0.083 0.083 0.057 0.055 0.053	0.057 0.158 0.007 0.649 0.016 0.016 0.029 0.028 0.028 0.038 0.038	0.085 0.127 0.053 0.672 1.109 0.042 0.016 0.026 0.009 0.001
ported 1 ported 1 le factor tes the index). 1985 as formanc	0.098 0.199	-0.011 0.344 0.649 0.088 0.088 0.058 0.056 0.057 0.052	$\begin{array}{c} 0.060\\ 0.158\\ 0.010\\ 0.381\\ 0.697\\ 0.032\\ 0.032\\ 0.0339\\ 0.0339\\ 0.032\\ 0.032\\ 0.032\\ 0.031\\ 0.031\end{array}$	0.084 0.122 0.054 0.689 1.154 0.017 0.017 0.011 0.013 0.003
Qirical pe (Q) is re n the on the or W denc market ecember The per	0.100 0.198	-0.007 0.358 0.694 0.091 0.003 0.063 0.064 0.061 0.061	$\begin{array}{c} 0.061\\ 0.158\\ 0.011\\ 0.386\\ 0.717\\ 0.717\\ 0.033\\ 0.041\\ 0.033\\ 0.041\\ 0.029\\ 0.040\\ 0.032\\ 0.032 \end{array}$	0.096 0.118 0.068 0.817 1.394 0.056 0.031 0.031 0.031 0.027 0.027
zed emp aure (CE based o where E weighted 668 to D (2001).	Periods 0.092 0.203	-0.020 0.306 0.595 0.083 0.082 0.059 0.059 0.057 0.057 0.054	$\begin{array}{c} 0.059\\ 0.160\\ 0.008\\ 0.368\\ 0.680\\ 0.680\\ 0.016\\ 0.016\\ 0.038\\ 0.033\\ 0.033\\ 0.033\\ 0.033\\ 0.031\\ 0.032\\ 0.$	Y Perio 0.096 0.107 0.107 0.073 0.058 0.058 0.058 0.032 0.032 0.032 0.032 0.032 0.032 0.032 0.027 0.027 0.025
annuali ant measure neasure rackets, value v April 19 ampbell	ecession 0.084 0.207	-0.031 0.265 0.525 0.074 0.072 0.057 0.055 0.055 0.051 0.046	0.056 0.161 0.161 0.044 0.0448 0.048 0.011 0.011 0.022 0.023 0.033 0.023 0.033 0.024 0.024 0.033 0.024 0.022 0.033 0.024 0.033 0.024 0.033 0.024 0.033 0.024 0.033 0.024 0.034 0.025 0.034 0.027 0.033 0.026 0.034 0.027 0.033 0.026 0.033 0.026 0.033 0.026 0.033 0.026 0.033 0.026 0.033 0.026 0.033 0.026 0.033 0.026 0.033 0.026 0.033 0.026 0.026 0.033 0.026 0.026 0.026 0.033 0.026	Volatilit 0.103 0.111 0.078 0.078 0.078 0.078 0.078 0.078 0.078 0.044 0.031 0.031 0.026 0.026 0.025
Subperiods The table reports the The certainty equival 1 Factor" the alpha r alpha is reported in b (proxied by the CRSF volatility period from 1997 as defined by Ca	Panel I: U.S. R Mean Std. Dev.	CEQ Sharpe Ratio Sortino Ratio I Factor Alpha (EW) I Factor Alpha (EW) 3 Factor Alpha (EW) 3 Factor Alpha (EW) 4 Factor Alpha (EW) 4 Factor Alpha (EW)	Mean Std. Dev. CEQ Sharpe Ratio Sortino Ratio I Factor Alpha (EW) 3 Factor Alpha (EW) 3 Factor Alpha (EW) 4 Factor Alpha (EW) 4 Factor Alpha (EW)	Panel III: HighMeanStd. Dev.CEQStratpe RatioSartino RatioAlpha 1 Factor (EW)Alpha 3 Factor (EW)Alpha 3 Factor (EW)Alpha 4 Factor (EW)Alpha 4 Factor (EW)Alpha 4 Factor (EW)

Table 5: Empirical performance metrics for the constrained minimum-variance portfolios with 3 months revision frequency over the selected

Table 0: Pe selected sub	ertorma	nce mei s with s	tric inf(a value	erence weio'hi	tor the ted hen	constr chmarl	alned fi	Inminio	m-varia	unce po	rtiollos	with 3	mont	ıs port.	Iolio re	VISION I	requen	cy over	. the
The table report have in the port that the particuties the one sided hy	ts the box folio) and ilar perfor ypotheses	otstrapped the value rmance me that the h	l p-values weighted etric of th bootstrapp	of the per benchme ie benchr ped alpha	erformance urk portfol nark port a t-value e	e metric c io. The p folio exce xceeds th	lifferences -values for eds that ϵ ie empiric	between ¹ : the certa of the resp al t-value	the constr unty equiv pective co and thus	valent mes nastrained indicates	aimum-val asure (CE minimum that the	riance poi Q), Sharp 1-variance generated	ttfolios (w e and Son e portfolio l alpha is	vith the d rtino ratic . P-value due to s	ifferent main sare com s for the ampling va	aximum w puted for alpha me ariation.	veight a si the one s asures ar The empl	ngle stoch ided hypo e compute oyed boot	κ may thesis ed for tstrap
methodologies a The difference in alpha measure b II for the low vc 1997 as defined	nre outline n the CE(based on t. batility pe by Campl	e in detail Q is report he one fac eriod from bell (2001)	in section ted for a r tor model (April 196	1 2.3. risk avers l, while ". 38 to Dec	ion coeffic Alpha 3 Fi ember 198	ient of γ actor" and 35 as defin	= 4. Alph 1 "Alpha [.] 1ed by Ca	ta measure 4 Factor" mpbell (2	es are con do so for 1 001), whil	aputed for the three le panel II	the descr and four f II reports	ribed fact factor moo the result	or models del. Pane is for the	s in sectio 1 I reports 10w volat	n 2.2. Her s the result ility period	eby deno ts for U.S d from Ja	tes "Alph . recession nuary 199	a 1 Factoi 1 periods, 86 to Dece	r" the panel ember
									Maxim	um Weigh	hts								
	2%	3%	4%	5%	6%	7%	8%	6%	10%	11%	12%	13%	14%	15%	16%	17%	18%	19%	20%
Panel I: I	U.S. Reces	ssion Perio	ds																
Δ Mean	0.015	0.010	0.012	0.024	0.050	0.048	0.067	0.061	0.119	0.123	0.129	0.125	0.106	0.144	0.140	0.143	0.006	0.113	0.104
∆ Std. Dev.	0.895	0.944	0.987	0.987	0.974	0.898	0.959	0.740	0.684	0.589	0.534	0.713	0.259	0.160	0.196	0.104	0.102	0.130	0.143
∆ CEQ ^ Sharne Ratio	0.004	0.006	0.008	0.011	0.022	0.031	0.039	0.057 0.069	0.099	0.123	0.125	0.096	0.137	0.190	0.166	0.185 0.137	0.012	0.203	0.242
Δ Sortino Ratio	0.014	0.014	0.016	0.027	0.049	0.051	0.068	0.064	0.115	0.124	0.126	0.134	0.098	0.128	0.123	0.133	0.008	0.103	0.091
Alpha 1 Factor	0.223	0.219	0.192	0.189	0.232	0.227	0.296	0.242	0.360	0.358	0.352	0.326	0.287	0.331	0.325	0.324	0.026	0.228	0.189
Alpha 3 Factor	0.029	0.058	0.034	0.054	0.096	0.096	0.097	0.138	0.222	0.202	0.222	0.246	0.152	0.176	0.191	0.208	0.030	0.142	0.150
Alpha 4 Factor	0.068	0.099	0.084	0.120	0.169	0.151	0.179	0.228	0.342	0.338	0.390	0.383	0.248	0.293	0.342	0.348	0.067	0.234	0.216
Panel II:	Low Vola	tility Perio	po																
Δ Mean	0.008	0.013	0.011	0.015	0.053	0.051	0.101	0.032	0.036	0.129	0.059	0.081	0.088	0.150	0.151	0.050	0.016	0.082	0.010
Δ Std. Dev.	0.710	0.820	0.885	0.914	0.902	0.796	0.867	0.653	0.472	0.759	0.612	0.499	0.382	0.345	0.428	0.314	0.405	0.095	0.016
ΔCEQ	0.005	0.009	0.005	0.009	0.037	0.048	0.079	0.023	0.044	0.107	0.068	0.092	0.104	0.191	0.167	0.078	0.014	0.159	0.045
Δ Sharpe Ratio	0.007	0.010	0.006	0.012	0.043	0.047	0.089	0.027	0.040	0.114	0.058	0.084	0.094	0.162	0.153	0.062	0.014	0.101	0.022
△ Sortino Ratio	0.007	0.011	0.005	0.012	0.050	0.051	0.098	0.029	0.048	0.143	0.076	0.094	0.120	0.186	0.192	0.086	0.016	0.114	0.020
Alpha 3 Factor	0.015	0.017	0.010	0.010	0.010	0.012	0.008	0.001	0.002	0.021	0.005	0.008	0.009	0.016	0.024	0.007	0.000	0.012	0.001
Alpha 4 Factor	0.035	0.024	0.017	0.015	0.017	0.011	0.011	0.003	0.003	0.013	0.009	0.006	0.004	0.020	0.020	0.003	0.004	0.001	0.000
Panel III:	High Vol	atility Peri	iod																
Δ Mean	0.511	0.632	0.618	0.775	0.764	0.672	0.562	0.817	0.674	0.510	0.633	0.718	0.566	0.521	0.572	0.624	0.961	0.854	0.582
Δ Std. Dev.	1.000	1.000	1.000	1.000	0.972	0.911	0.938	0.793	0.565	0.318	0.399	0.409	0.206	0.185	0.248	0.285	0.163	0.191	0.132
Δ CEQ	0.189	0.328	0.420	0.617	0.632	0.590	0.456	0.756	0.661	0.542	0.660	0.722	0.638	0.621	0.630	0.664	0.966	0.878	0.686
△ Sharpe Ratio	0.047	0.131	0.317	0.572	0.608	0.585	0.434	0.760	0.678	0.572	0.674	0.733	0.662	0.642	0.649	0.684	0.965	0.880	0.706
Δ Sortino Ratio	0.059	0.115	0.248	0.495	0.552	0.503	0.351	0.717	0.578	0.452	0.560	0.641	0.555	0.531	0.573	0.599	0.948	0.842	0.604
Alpha 1 Factor	0.000	0.000	0.010	0.033	0.066	0.045	0.030	0.134	0.074	0.038	0.067	0.102	0.043	0.047	0.049	0.060	0.440	0.187	0.036
Alpha 3 Factor	0.007	0.018	0.033	0.127	0.153	0.169	0.073	0.366	0.228	0.140	0.212	0.240	0.163	0.152	0.127	0.133	0.636	0.284	0.117
АІрпа 4 гастог	U.U40	0.040	0.00	0.233	000.0	0.430	0.401	0.114	100.0	0.231	0.420	Tec.U	0.004	ATC.U	0.200	0.212	eno.U	eee.U	1.62.0

4 J ÷ J.L. 2 c 4+: ÷ J J • 5 ¢ F ς Tabl

31

Table /: Fe selected sub	rtormai	nce met 2 with 9	UTIC INIG	erence Ilv wei	ion the	CONSUR	ained n ark	Inminic	n-varıa	nce poi	rtiolios	WILD 3	montf	is porti	ollo rev	/ISION I	requen	cy over	the
The table report have in the port that the particu the one sided hy	ts the boo folio) and lar perfor potheses	the equal; mance me that the k	I p-values y weighter etric of th pootstrapt	of the pe d benchm e benchr oed alphe	rformance tark portfo nark portfort nark portfort	t metric d blio. The I colio excet xceeds th	ifferences -values fc ds that o e empirica	between t or the cert: of the resp al t-value	the constra ainty equi bective con and thus	ained min ivalent me nstrained indicates	nimum-var assure (CE minimum that the	riance por JQ), Sharj 1-variance generated	tfolios (w pe and So portfolio alpha is	ith the di rtino ratic . P-value due to sa	fferent ma s are com s for the mpling va	wimum w puted for alpha me vriation.	reight a si the one s asures ar The empl	ngle stoch ided hypo e compute oyed boot	t may thesis ed for sstrap
The uncouncies a The difference in alpha measure b II for the low vo 1997 as defined	a the CEC ased on the latility pe by Campt	lin decau lis report he one fact briod from coll (2001)	in section ced for a r tor model, April 196).	isk avers , while ", 8 to Dec	ion coeffic. Alpha 3 F ² ember 195	ient of γ = actor" and 55 as defin	= 4. Alph "Alpha 4 ed by Cau	a measure 4 Factor" (mpbell (2	ss are com do so for t 301), whil	puted for the three a e panel II	the descr and four f I reports	ribed facto actor moo the result	or models del. Panel s for the	in section I reports low volati	a 2.2. Her the result lity perioc	eby denot s for U.S. l from Ja	tes "Alph . recessior nuary 198	a 1 Factor t periods, 86 to Dece	," the panel ember
									Maxim	um Weigh	nts								
	2%	3%	4%	5%	6%	7%	8%	6%	10%	11%	12%	13%	14%	15%	16%	17%	18%	19%	20%
Panel I: (J.S. Reces	sion Perio	ds																
Δ Mean	0.153	0.143	0.138	0.148	0.166	0.157	0.177	0.162	0.215	0.218	0.220	0.222	0.192	0.213	0.208	0.211	0.036	0.186	0.165
∆ Std. Dev.	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	0.999	0.998	0.996	0.998	0.966	0.944	0.986	0.981	0.989	0.894	0.826
A CEQ	0.009	0.007	0.011	0.008	0.023	0.029	0.024	0.028	0.049	0.052	0.065	0.033	0.063	0.080	0.074	0.079	0.011	0.082	0.095
△ Snarpe Katio ∧ Sortino Batio	0.176	0.152	0.139	0.136	0.164	0.170	0.177	0.160	0.221	0.240	0.234	0.231	0.192	0.224	0.220	0.224	0.031	0.187	0.165
Alpha 1 Factor	0.188	0.188	0.186	0.184	0.230	0.222	0.284	0.245	0.351	0.345	0.337	0.325	0.299	0.325	0.334	0.323	0.023	0.227	0.200
Alpha 3 Factor	0.029	0.036	0.024	0.047	0.088	0.084	0.094	0.130	0.209	0.199	0.214	0.236	0.145	0.168	0.190	0.202	0.025	0.144	0.135
Alpha 4 Factor	0.067	0.079	0.063	0.106	0.165	0.144	0.174	0.215	0.336	0.333	0.384	0.370	0.256	0.290	0.338	0.352	0.068	0.228	0.212
Panel II:	Low Volat	tility Peric	pc																
Δ Mean	0.739	0.690	0.662	0.678	0.706	0.689	0.713	0.619	0.640	0.740	0.691	0.662	0.691	0.745	0.740	0.652	0.530	0.661	0.427
Δ Std. Dev.	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	0.999	0.987
Δ CEQ	0.199	0.165	0.161	0.167	0.215	0.216	0.233	0.177	0.201	0.257	0.232	0.232	0.255	0.312	0.288	0.238	0.128	0.282	0.173
∆ Sharpe Ratio	0.485	0.436	0.391	0.395	0.472	0.465	0.499	0.374	0.414	0.535	0.465	0.445	0.507	0.591	0.559	0.451	0.264	0.505	0.285
∆ Sortino Ratio	0.463	0.427	0.384	0.406	0.494	0.487	0.521	0.380	0.443	0.567	0.501	0.464	0.540	0.612	0.596 0.2 <u>9</u> 6	0.494 0.1 	0.274	0.507	0.277
Alpha 1 Factor Alpha 3 Factor	0.018	0.020	0.007	0.008	0.011	0.014	0.008	0.006	0.006	0.028	0.008	0.012	0.007	0.020	0.025	0.008	0.000	0.012	0.000
Alpha 4 Factor	0.028	0.023	0.015	0.017	0.016	0.014	0.012	0.006	0.004	0.020	0.013	0.011	0.006	0.032	0.023	0.002	0.005	0.004	0.000
Panel III:	High Vola	atility Peri	iod																
Δ Mean	0.313	0.405	0.393	0.541	0.540	0.482	0.394	0.621	0.492	0.372	0.476	0.556	0.418	0.391	0.433	0.474	0.882	0.745	0.456
Δ Std. Dev.	1.000	1.000	1.000	1.000	1.000	0.990	0.995	0.971	0.908	0.771	0.841	0.820	0.663	0.612	0.654	0.673	0.498	0.532	0.439
ΔCEQ	0.102	0.159	0.197	0.306	0.313	0.322	0.240	0.472	0.392	0.303	0.402	0.480	0.380	0.354	0.395	0.423	0.860	0.727	0.467
Δ Sharpe Ratio	0.025	0.061	0.140	0.294	0.304	0.312	0.222	0.487	0.417	0.320	0.426	0.502	0.401	0.380	0.414	0.443	0.875	0.745	0.482
Δ Sortino Ratio	0.030	0.066	0.124	0.251	0.273	0.272	0.189	0.454	0.376	0.251	0.348	0.425	0.350	0.340	0.369	0.397	0.849	0.686	0.420
Alpha 1 Factor	0.000	0.001	0.009	0.035	0.056	0.047	0.021	0.116	0.070	0.043	0.079	0.096	0.058	0.050	0.052	0.046	0.436	0.178	0.028
Alpha 3 Factor	0.005	0.011	0.022	0.107	0.139	0.149	0.071	0.334	0.212	0.113	0.209	0.220	0.155	0.142	0.120	0.121	0.621	0.270	0.106
Alpha 4 Factor	0.024	0.034	0.064	0.280	0.310	0.399	0.251	0.685	0.515	0.272	0.406	0.314	0.344	0.294	0.260	0.261	0.586	0.317	0.239

-¢ ÷ ÷ . f 1 2 E

32

The table reports su a single stock may <i>l</i> returns are normally section 2.2. Hereby market factor bench VW the value weigh II does so for the two	ummariz nave in t distrib denotes mark for mark for ted mar elve mor	ing portf he portf uted. Th "Alpha r every a ket benc ths port	iolio stat olio. Tha ne certai 1 Factor Ipha is r hmark p folio rev	istics as e summa mty equi r" the al eported bortfolio ision fre	well as ury statii valent n pha mes in brack (proxied quency.	the ann stics sho neasure - asure bas ets, whe ets, whe The per	w the m w the m (CEQ) is sed on th re EW d CRSP vi formance	mpirical ean num s reporte ne one fa enotes tl alue weig metrics	perform ber of st d for a 1 ctor mo ctor mo ne equal for both	ance me cocks in cisk aver del, whil ly weight urket ind 1 benchm	trics for the port sion coel le "Alph ted marh ted marh ex). Par nark por	the cons folio as v ficient o a 3 Fact tet bench tet I rep tfolios a	trained well as the vell as the $f \gamma = 4$. or" and or" and mark points the points the re, for the red	minimul he Jarqu Alpha "Alpha" ortfolio ortfolio results f he purpo	n-varian le-Berra measure 4 Facto proxied or the si se of vis	ce portf test p-v s are col by the (by the (-a-vis pe	olios wit alues fo mputed for the CRSP ec CRSP ec rs portformar	th the r r the hy for the three a qually w plio revii nce com	espective r pothesis t described m four fa eighted m sion freque parison, re	hat the pc factor mo ctor mode arket inde mcy, while ported as	weight artfolio dels in l. The x) and vell.
	2%	3%	4%	5%	8%	7%	8%	%6	10%	11%	12%	13%	14%	15%	16%	17%	18%	19%	20%		EW
Panel I: 6 months	adjustme	ent freque	ncy																		
Mean no. of Stocks	53	38	29	25	22	20	18	17	16	16	15	14	14	14	13	13	12	12	12		
Jarque-Berra p-values	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0000.0	0.000	0.000	0000.0	0.000	0.000	0.000	0.000
Mean	0.070	0.072	0.070	0.067	0.074	0.081	0.086	060.0	0.096	0.091	0.100	0.107	0.092 (0.094	0.095 (0.100 0	0.101 (0.094	0.106	0.056	0.144
Std. Dev.	0.132	0.131	0.134	0.139	0.140	0.145	0.150	0.154	0.159	0.161	0.164	0.168	0.171 (0.174	0.174 0	0.179 0).183 (0.184	0.193	0.156	0.198
CEQ	0.035	0.038	0.034	0.028	0.034	0.039	0.041	0.042	0.046	0.039	0.046	0.051	0.034 (0.033	0.035 ().036 (0.034 (0.026	0.032	0.008	0.065
Sharpe Ratio	0.530	0.555	0.523	0.482	0.526	0.557	0.571	0.583	0.605	0.563	0.609	0.639	0.538 (0.539	0.548 (.557 (.550 (0.509	0.551	0.362	0.725
Sortino Ratio	0.929	0.984	0.949	0.867	0.944	1.006	1.044	1.085	1.132	1.053	1.170	1.221	1.017	1.024	1.033 1	.044 1	1.022 (0.943	1.025	0.514	1.056
Alpha 1 Factor (EW)	0.030	0.035	0.034	0.031	0.038	0.046	0.050	0.055	0.061	0.056	0.066	0.074	0.058 (0.059	0.062 (.066 (.068 (0.060	0.074		
Alpha 1 Factor (VW)	0.030	0.034	0.033	0.030	0.037	0.045	0.050	0.054	0.060	0.055	0.065	0.074	0.056 (0.058	0.061 (0.064 0	.067 (0.060	0.071		
Alpha 3 Factor (EW)	0.025	0.031	0.031	0.029	0.037	0.046	0.049	0.055	0.062	0.057	0.068	0.073	0.059 (0.062	0.063 (0.068 (0.069 (0.060	0.078		
Alpha 3 Factor (VW)	0.013	0.020	0.020	0.000 81U.U	0.027	0.035	0.040	0.045	10.051	0.047	0.058	0.065	J. U5U (0.052	0.054 0	0.058	0.059 (1.051	0.067		
Alpha 4 Factor (EW)	0.000	\$10.0	110.0	0.006	0.015	0.024	0.027	0.021	0.035	160.0	0.041	0.059	10.02	0.034	0.000	0.039	0.042	0.030	0.051		
Alpha 4 Factor (V W) Douel II, 12 month	u.uuo adiiiatw	0.014	710.0	000.0	e10.0	0.024	170.0	100.0	100.0	ee0.0	U.U41	700.0	400.r	, eu.,	- Acu.). 044	800.0	100.0		
Mean no. of Stocks	53	38	30	25	23	21	18	17	17	16	16	13	15	15	14	13	12	13	12		
Jarque-Berra p-values	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0000) 000 (0.000	0.000	0.000	0.000
Mean	0.066	0.066	0.068	0.064	0.070	0.073	0.077	0.083	0.089	0.093	0.093	0.100	0.099 (0.107	0.105 (0.111 0	0.111 (0.104	0.114	0.056	0.144
Std. Dev.	0.133	0.130	0.132	0.138	0.139	0.144	0.147	0.150	0.156	0.158	0.158	0.161	0.162 (0.165	0.167 0	0.170 0).169 (0.169	0.174	0.156	0.198
CEQ	0.031	0.032	0.033	0.027	0.031	0.031	0.034	0.038	0.040	0.043	0.043	0.048	0.047	0.052	0.049 (0.054 0).053 (0.047	0.053	0.008	0.065
Sharpe Ratio	0.500	0.509	0.513	0.469	0.503	0.506	0.522	0.554	0.571	0.589	0.585	0.622	0.612 (0.645	0.629 (.655 ().654 (0.617	0.654	0.362	0.725
Sortino Ratio	0.877	0.888	0.916	0.843	0.907	0.914	0.930	1.011	1.046	1.094	1.093	1.139	1.149	1.235	1.190	243 1	1.240	1.174	1.230	0.514	1.056
Alpha I Factor (EW)	0.026	0.027	0.030	0.027	0.034	0.036	0.042	0.049	0.054	0.058	0.059	0.069	J.U65 (0.072	0.072 (0.078	0.080	0.072	0.081		
Alpha I Factor (VW)	0.026	0.022	0.030	0.022	0.033	0.035	0.040	0.047	0.053	0.052	0.050	0.071	J.U63 (07.0.0	0.070	0.076	0.059	070.0	0.080		
Alpha 3 Factor (UW)	0.008	0.013	0.015	0.011	0.018	0.021	0.026	0.035	0.041	0.047	0.048	0.060	0.054 (0.062	0.063	020.000.000.000.0000.000000000000000000	0.072 (0.062	0.072		
Alpha 4 Factor (EW)	0.006	0.008	0.011	0.004	0.009	0.011	0.011	0.022	0.025	0.032	0.031	0.041	0.033 (0.039	0.041 0	0.048 (0.052 (0.042	0.054		
Alpha 4 Factor (VW)	0.007	0.010	0.012	0.004	0.010	0.012	0.013	0.023	0.026	0.034	0.032	0.040	0.035 (0.042	0.043 (0.050 (0.054 (0.045	0.056		

Table 8: Empirical performance metrics for the constrained minimum-variance portfolios with 6 and 12 months portfolio revision frequency

the	
over	
frequency	
revision f	
portfolio	
6 months	
s with	
portfolios	
-variance	
minimum	
nstrained	
the co	
nce for	er 2007
infere	ecembe
e metric	38 to D
ormance	vpril 190
Perfc	m A
9: F	fro
) elc	iod
Tal	per

Sortino ratio and certainty equivalents; inference concerning the different alpha measures is based on an residual bootstrap. Both bootstrap methodologies are outline in detail in section 2.3. The "Alpha 1 Factor" the alpha measure based on the one factor model, while "Alpha 3 Factor" and "Alpha 4 Factor" do so for the three and four factor model. The market factor benchmark for every alpha is reported in brackets, where EW denotes the equally weighted market benchmark (proxied by the CRSP equally weighted market index) and VW the value weighted market benchmark (proxied by the CRSP equally weighted market index). Panel I reports the results for the value weighted benchmark (proxied by the CRSP value weighted market index). Panel I reports the results for the value weighted benchmark portfolio, while panel II does so for the equally weighted benchmark (proxied by the CRSP value weighted market index). The table reports the bootstrapped p-values for the performance metrics for each constrained minimum-variance portfolio with the respective maximum weight a single stock may have in the portfolio over the complete sample period from April 1968 to December 2007. We use an observation bootstrap for the statistical performance metric inference concerning the Sharpe ratio, difference in certainty equivalent measures (CEQ) is reported for a risk aversion coefficient of $\gamma = 4$. Alpha measures are computed for the described factor models in section 2.2. Hereby denotes portfolio.

									Maxin	num Weigh	nts								
	2%	3%	4%	5%	6%	7%	8%	6%	10%	11%	12%	13%	14%	15%	16%	17%	18%	19%	20%
Panel I: V.	alue weighte	d benchma	urk																
Δ Mean	0.156	0.133	0.204	0.293	0.168	0.087	0.068	0.050	0.030	0.062	0.026	0.011	0.063	0.052	0.049	0.037	0.046	0.077	0.035
Δ Std. Dev.	1.000	1.000	1.000	1.000	1.000	0.983	0.866	0.663	0.376	0.283	0.205	0.109	0.060	0.031	0.035	0.006	0.003	0.002	0.001
ΔCEQ	0.030	0.023	0.052	0.116	0.064	0.043	0.048	0.044	0.033	0.082	0.036	0.027	0.127	0.133	0.129	0.119	0.156	0.245	0.184
Δ Sharpe Ratio	0.030	0.025	0.068	0.145	0.076	0.048	0.049	0.045	0.031	0.080	0.033	0.022	0.105	0.098	0.094	0.090	0.096	0.180	0.115
Δ Sortino Ratio	0.024	0.017	0.046	0.099	0.054	0.034	0.037	0.031	0.023	0.053	0.022	0.013	0.069	0.067	0.073	0.065	0.078	0.137	0.088
Alpha 1 Factor	0.013	0.008	0.016	0.034	0.015	0.008	0.005	0.003	0.002	0.009	0.001	0.000	0.007	0.007	0.005	0.005	0.007	0.010	0.004
Alpha 3 Factor	0.016	0.010	0.018	0.027	0.011	0.004	0.003	0.000	0.000	0.003	0.000	0.000	0.002	0.002	0.002	0.001	0.002	0.005	0.001
Alpha 4 Factor	0.302	0.192	0.224	0.329	0.200	0.097	0.090	0.067	0.044	0.050	0.031	0.017	0.070	0.060	0.045	0.036	0.042	0.065	0.015
Panel II: Eq	jually weigh:	ted benchn	nark																
Δ Mean	0.789	0.751	0.786	0.816	0.725	0.609	0.525	0.461	0.374	0.461	0.335	0.243	0.431	0.407	0.382	0.329	0.326	0.406	0.285
Δ Std. Dev.	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	0.997	0.989	0.975	0.968	0.942	0.874	0.861	0.680
ΔCEQ	0.107	0.090	0.134	0.201	0.135	0.104	0.096	0.092	0.078	0.126	0.083	0.063	0.178	0.187	0.165	0.163	0.179	0.268	0.226
Δ Sharpe Ratio	0.201	0.156	0.254	0.374	0.243	0.183	0.166	0.149	0.116	0.191	0.118	0.084	0.249	0.246	0.220	0.206	0.220	0.317	0.239
Δ Sortino Ratio	0.217	0.161	0.227	0.342	0.240	0.176	0.158	0.134	0.107	0.164	0.094	0.070	0.205	0.207	0.194	0.180	0.203	0.279	0.230
Alpha 1 Factor	0.014	0.009	0.014	0.024	0.012	0.007	0.004	0.003	0.000	0.009	0.000	0.000	0.005	0.006	0.007	0.006	0.006	0.010	0.003
Alpha 3 Factor	0.019	0.012	0.021	0.029	0.012	0.001	0.001	0.001	0.000	0.002	0.000	0.000	0.001	0.002	0.002	0.000	0.002	0.004	0.001
Alpha 4 Factor	0.304	0.190	0.223	0.333	0.195	0.088	0.096	0.063	0.047	0.061	0.031	0.013	0.072	0.074	0.059	0.040	0.041	0.070	0.021

/er	
y 01	
enc	
onbe	
۱ fre	
sior	
evi	
io 1	
tfol	
por	
hs	
lon	
2 m	
h 1	
wit	
ios	
tfol	
por	
lce	
riar	
l-Va	
unu	
nin	
E	
ned	
trai	
ons	
le c	
r tł	20C
e fo	1 2(
ence	nbe
lfer	ecei
c ir	Õ
etri	8 tc
e m	196
anc	oril
rm	AF
erfc	rom
Ч.	йĥ
10	erio
$_{\rm ble}$	e p
Ĥ	$_{\mathrm{th}}$

over the complete sample period from April 1968 to December 2007. The certainty equivalent measure (CEQ) is reported for a risk aversion coefficient of $\gamma = 4$. Alpha measures are computed for the described factor models in section 2.2. Hereby denotes "Alpha 1 Factor" the alpha measure based on the one factor model, while "Alpha 3 Factor" do so for the three and four factor model. The market factor benchmark for every alpha is reported in brackets, where EW denotes the equally weighted market benchmark (proxied by the CRSP equally weighted market index) and VW the value weighted market benchmark (proxied by the CRSP value weighted market index). Panel I reports the results for the value weighted benchmark portfolio, while The table reports the annualized empirical performance metrics for the constrained minimum-variance portfolios with the respective maximum weight a single stock may have in the portfolio panel II does so for the equally weighted benchmark portfolio.

									Maxim	um Weigh	tts								
	2%	3%	4%	5%	8%	7%	8%	6%	10%	11%	12%	13%	14%	15%	16%	17%	18%	19%	20%
Danal I. V.	alua mai <i>w</i> hte	smdanad be																	
Δ Mean	0.240	0.275	0.273	0.375	0.245	0.188	0.153	0.099	0.059	0.048	0.053	0.039	0.034	0.014	0.020	0.014	0.017	0.028	0.010
Δ Std. Dev.	1.000	1.000	1.000	1.000	1.000	0.996	0.982	0.908	0.473	0.394	0.373	0.228	0.234	0.100	0.083	0.042	0.049	0.063	0.019
ΔCEQ	0.053	0.059	0.072	0.175	0.100	0.103	0.094	0.071	0.061	0.051	0.058	0.054	0.047	0.029	0.043	0.038	0.041	0.064	0.039
Δ Sharpe Ratio	0.069	0.076	0.093	0.211	0.134	0.114	0.109	0.082	0.060	0.053	0.054	0.049	0.046	0.026	0.042	0.029	0.032	0.055	0.031
Δ Sortino Ratio	0.049	0.056	0.066	0.161	0.096	0.082	0.078	0.050	0.046	0.034	0.040	0.040	0.029	0.016	0.027	0.022	0.027	0.039	0.021
Alpha 1 Factor	0.025	0.032	0.023	0.042	0.022	0.016	0.011	0.007	0.007	0.001	0.002	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
Alpha 3 Factor	0.060	0.038	0.033	0.055	0.023	0.021	0.014	0.006	0.005	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
Alpha 4 Factor	0.293	0.243	0.217	0.317	0.247	0.233	0.251	0.113	0.101	0.055	0.064	0.038	0.059	0.040	0.029	0.011	0.008	0.022	0.005
Panel II: Ec	qually weigh	ited benchr.	nark																
Δ Mean	0.818	0.821	0.799	0.833	0.775	0.727	0.657	0.582	0.488	0.431	0.444	0.364	0.360	0.276	0.301	0.229	0.237	0.302	0.217
Δ Std. Dev.	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	0.994
ΔCEQ	0.146	0.142	0.147	0.237	0.179	0.180	0.152	0.122	0.117	0.094	0.106	0.079	0.085	0.071	0.083	0.066	0.074	0.108	0.081
Δ Sharpe Ratio	0.318	0.296	0.300	0.442	0.341	0.318	0.299	0.224	0.191	0.160	0.189	0.131	0.139	0.100	0.122	0.104	0.094	0.152	0.112
Δ Sortino Ratio	0.315	0.316	0.293	0.403	0.315	0.298	0.287	0.207	0.183	0.145	0.159	0.120	0.113	0.081	0.114	0.095	0.092	0.134	0.111
Alpha 1 Factor	0.038	0.034	0.030	0.050	0.022	0.019	0.014	0.008	0.007	0.001	0.001	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
Alpha 3 Factor	0.054	0.038	0.037	0.053	0.023	0.020	0.015	0.004	0.004	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
Alpha 4 Factor	0.294	0.243	0.205	0.325	0.252	0.236	0.231	0.112	0.087	0.051	0.052	0.031	0.051	0.035	0.025	0.012	0.009	0.019	0.005

References

- BAKER, N., AND R. HAUGEN (1991): "The Efficient Market Inefficiency of Capitalization-Weighted Stock Portfolios," *Journal of Portfolio Management*, 17(1), 35–40.
- BALI, T. G., N. CAKICI, X. YAN, AND Z. ZHANG (2005): "Does Idiosyncratic Risk Really Matter?," *Journal of Finance*, 60, 905–929.
- BLACK, F., AND R. LITTERMAN (1992): "Global Portfolio Optimization," *Financial* Analysts Journal, 48, 28–43.
- BLOOMFIELD, T., R. LEFTWICH, AND J. LONG (1977): "Portfolio Strategies and Performance," *Journal of Financial Economics*, 5, 201–218.
- BUTLER, A. W., G. GRULLON, AND J. P. WESTON (2001): "Can Managers Forecast Aggregate Market Returns?," *Journal of Finance*, 60, 963–986.
- CAMPBELL, J. Y., M. LETTAU, B. G. MALKIEL, AND Y. XU (2001): "Have individual stocks become more volatile? An empirical exploration of idiosyncratic risk," *Journal of Finance*, 56, 1–43.
- CARHART, M. (1997): "On the Persistence in Mutual Fund Performance," Journal of Finance, 52, 57–82.
- CHAN, L. K. C., J. KARCESKI, AND J. LAKONISHOK (1999): "On Portfolio Optimization: Forecasting Covariances and Choosing the Risk Model," *The Review of Financial Studies*, 12, 937–974.
- CHOPRA, V. K. (1993): "Improving Optimization," Journal of Investing, 8, 51–59.
- CHOPRA, V. K., AND W. T. ZIEMBA (1993): "The Effects of Errors in Means, Variances, and Covariances on Optimal Portfolio Choice," *Journal of Portfolio Management*, 4, 6–11.
- CLARKE, R., H. D. S., AND S. THORLEY (2006): "Minimum-variance portfolios in the U.S. equity market," *Journal of Portfolio Management*, 33, 10–24.
- DEMIGUEL, V., L. GARLAPPI, AND R. UPPAL (2007): "Optimal Versus Naive Diversification: How Inefficient is the 1/N Portfolio Strategy?," *The Review of Financial Studies*, Forthcoming.
- DIMSON, E. (1979): "Risk Management when Shares are Subject to infrequent trading," Journal of Financial Economics, 7, 197–226.
- EFRON, B. (1979): "Bootstrap methods: Another look at the Jacknife," Annals of Statistics, 7, 1–26.
- ELTON, E. J., AND M. J. GRUBER (1997): "Modern Portfolio Theory, 1950 to date," *Journal of Banking and Finance*, 21, 1743–1759.
- FABOZZI, F. J., P. N. KOLM, D. A. PACHMANOVA, AND S. M. FOCARDI (2007): "Robust Portfolio Optimization," *Journal of Portfolio Management*, 33, 40–48.

- FAMA, E. F. (1965): "The Behavior of Stock-Market Prices," *Journal of Business*, 38, 34–105.
 - (1970): "Multiperiod consumption-investment decisions," American Economic Review, 60, 163–174.
- FAMA, E. F., AND K. R. FRENCH (1992): "The Cross-Section of Expected Stock Returns," Journal of Finance, 47, 427–465.
- FROST, P. A., AND J. E. SAVARINO (1986): "An Empirical Bayes Approach to Efficient Portfolio Selection," *Journal of Financial and Quantitative Analysis*, 21, 293–305.

(1988): "For Better Performance Constrain Portfolio Weights," Journal of Portfolio Management, 15, 29–34.

- GARLAPPI, L., R. UPPAL, AND T. WANG (2007): "Portfolio Selection with Parameter and Model Uncertainty: a Multi-prior Approach," *The Review of Financial Studies*, 20, 41–80.
- GOYAL, A., AND P. SANTA-CLARA (2003b): "Idiosyncratic Risk Matters," *Journal of Finance*, 58, 975–1007.
- GOYAL, A., AND I. WELCH (2003a): "Predicting the Equity Premium with Dividend Ratios," *Management Science*, 49, 639–654.

(2007): "A Comprehensive Look at the Empirical Performance of Equity Premium Prediction," *The Review of Financial Studies*, Forthcoming.

- GREEN, R., AND B. HOLLIFIELD (1992): "When will mean-variance efficient portfolios be well diversified," *Journal of Finance*, 47, 1785–1809.
- HAKANSSON, N. (1970): "Optimal investment and consumption strategies under risk for a class of utility functions," *Econometrica*, 38, 587–607.
 - (1974): "Convergence in multiperiod portfolio choice," *Journal of Financial Economics*, 1, 201–224.
- JAGANNATHAN, R., AND T. MA (2003): "Risk Reduction in Large Portfolios: Why Imposing the Wrong Constraints Helps," *Journal of Finance*, 58, 1651–1684.
- JAMES, W., AND C. STEIN (1961): "Estimation with quadratic loss," in *Proceedings* of the 4th Berkeley Symposium on Probability and Statistics, Berkeley. University of California Press.
- JENSEN, M. C. (1969): "Risk, The Pricing of Capital Assets and the Evaluation of Investment Portfolios," *Journal of Business*, 42, 167–247.
- JOBSON, J. D., AND B. KORKIE (1980): "Estimation for Markowitz Efficient Portfolios," Journal of the American Statistical Association, 75, 544–554.

(1981a): "Performance Hypothesis Testing with the Sharpe and Treynor Measures," *Journal of Finance*, 36, 889–908.

(1981b): "Putting Markowitz Theory at work," Journal of Portfolio Management, 7(4), 70–74.

- JOBSON, J. D., B. KORKIE, AND V. RATTI (1979): "Improved estimation for Markowitz portfolios using James-Stein type estimators," in *Proceedings of the American Statistical Association Business and Economics Statistics Section*, Washington. American Statistical Association.
- JORION, P. (1985): "International Portfolio Diversification with Estimation Risk," *Journal of Business*, 58, 259–278.
 - (1986): "Bayes-Stein Estimation for Portfolio Analysis," *Journal of Financial and Quantitative Analysis*, 21, 279–292.
- KOSOWSKI, R. (2004): "Do Mutual Funds Perform When It Matters Most to Investors? US Mutual Fund Performance and Risk in Recessions and Expansions," *Working Paper*.
- KOSOWSKI, R., N. Y. NAIK, AND M. TEOC (2007): "Do Hedge Funds Deliver Alpha? A Bayesian and Bootstrap Analysis," *Journal of Financial Economics*, Volume 84 (1), 229–264.
- LEDOIT, O., AND M. WOLF (2003): "Improved estimation of the covariance matrix of stock returns with an application to portfolio selection," *Journal of Empirical Finance*, 10, 603–621.
- (2008): "Robust Performance Hypothesis Testing with the Sharpe Ratio," *Journal of Empirical Finance*, Forthcoming.
- LEHMANN, B., AND D. MODEST (1987): "Mutual Fund Performance Evaluation: A Comparison of Benchmarks and Benchmark Comparisons," *Journal of Finance*, 42, 233–265.
- LEVINE, S. N. (1972): Financial Analyst's Handbook: Methods, Theory and Portfolio Management. Dow Jones-Irwin, Inc.
- MALKIEL, B. G., AND Y. XU (1997): "Risk and return revisited," Journal of Portfolio Management, 23, 9–14.
- (2004): "Idiosyncratic risk and security returns," Working Paper.
- MARKOWITZ, H. (1952): "Portfolio Selection," Journal of Finance, 7, 77–91.
- (1959): Portfolio Selection: Efficient Diversification of Investments. Wiley, New York.
- MCINISH, T., AND R. WOOD (1986): "Adjusting for Beta Bias: An Assessment of Alternative Techniques: A Note," *Journal of Finance*, 41, 277–286.
- MEMMEL, C. (2003): "Performance Hypothesis Testing with the Sharpe Ratio," *Finance Letters*, 1, 21–23.
- MERTON, R. C. (1980): "On Estimating the Expected Return on the Market: An Exploratory Investigation," *Journal of Financial Economics*, 8, 323–361.
 - (1990): Continuous Time Finance. Basil Blackwell, Oxford.
- MOSSIN, J. (1969): "Optimal multiperiod portfolio policies," *Journal of Business*, 41, 215–229.

- NAVIDI, W. (1989): "Edgeworth expansions for bootstrapping regression models," *The Annals of Statistics*, 17, 1472–1478.
- NELSON, D. (1992): "Filtering and forecasting with misspecified ARCH models: Getting the right variance with the wrong model," *Journal of Econometrics*, 52, 61–90.
- NEWEY, W. K., AND K. D. WEST (1987): "A simple, positive semi-definite, heteroskedasticity and autocorrelation consistent covariance matrix," *Econometrica*, 55, 703–708.
- PASTOR, L., AND R. STAMBAUGH (2003): "Liquidity Risk and Expected Stock Returns," Journal of Political Economy, 111, 642–685.
- SCHOLES, M., AND J. WILLIAMS (1977): "Estimating Betas from Nonsynchronous Data," *Journal of Financial Economics*, 5, 309–327.
- SHARPE, W. F. (1963): "A simplified model for portfolio analysis," *Management Science*, 9, 277–293.
- (1964): "Capital Asset Prices: A Theory of Market Equilibrium under Conditions of Risk," *Journal of Finance*, 19, 425–442.
- STEIN, C. (1956): "Inadmissibility of the Usual Estimator for the Mean of a Multivariate Normal Distribution," in *Proceedings of the 3rd Berkeley Symposium on Probability and Statistics*, Berkeley. University of California Press.
- TOBIN, J. (1958): "Liquidity Preference as Behavior Towards Risk," *Review of Economic Studies*, 25, 65–85.
- VINOD, H. D., AND M. R. MOREY (1999): "Confidence Intervals and Hypothesis Testing for the Sharpe and Treynor Performance Measures: A Bootstrap Approach," in *Computational Finance*, ed. by Y. S. Abu-Mostafa, B. LeBaron, A. W. Lo, and A. S. Weigand, pp. 25–39. The MIT Press, Cambridge, Massachusetts.